

Received 20 October 2023, accepted 1 November 2023, date of publication 8 November 2023, date of current version 14 November 2023.

Digital Object Identifier 10.1109/ACCESS.2023.3331078

# **RESEARCH ARTICLE**

# **Examining IoT-Based Smart Campus Adoption Model: An Investigation Using Two-Stage Analysis Comprising Structural Equation Modelling and Artificial Neural Network**

# RADHWAN SNEESL<sup>®1,2</sup>, YUSMADI YAH JUSOH<sup>®1</sup>, MARZANAH A. JABAR<sup>®1</sup>, AND SALFARINA ABDULLAH<sup>®1</sup>

<sup>1</sup>Department of Software Engineering and Information System, Faculty of Computer Science and Information Technology, Universiti Putra Malaysia (UPM), Serdang, Selangor 43400, Malaysia

<sup>2</sup>College of Science, University of Basrah, Basrah 61001, Iraq

Corresponding authors: Radhwan Sneesl (radhwan.sneesl@gmail.com) and Yusmadi Yah Jusoh (yusmadi@upm.edu.my)

This work was supported by the Ministry of Higher Education and Universiti Putra Malaysia for using the Fundamental Research Grant Scheme under Grant FRGS/1/2021/ICT01/UPM/02/2.

This work involved human subjects or animals in its research. Approval of all ethical and experimental procedures and protocols was granted by the Ethics Committee for Research Involving Human Subjects Universiti Putra Malaysia.

**ABSTRACT** The progress and evolution of technology have been rapidly transforming various aspects of our society and daily lives, including colleges and campuses into smarter environments compared to the past. Despite the numerous advantages offered by cutting-edge technologies, such as IoT-based smart campuses, academic research on their implementation suffers from a significant lack of comprehensive information necessary to deliver efficient smart campus solutions. Therefore, the focus of this study is to investigate the significance of IoT-based smart campus adoption from 14 proposed hypotheses. The researchers collected data from stakeholders affiliated with universities in Iraq, resulting in a dataset of 442 observations. To analyze the data, a two-stage approach was employed, consisting of structural equation modeling (SEM) and reevaluated with the artificial neural networks (ANN) method. The findings provide evidence supporting the significance of various constructs. In particular, the model demonstrates satisfactory predictive relevance, indicating its effectiveness in making accurate predictions or forecasts. The ANN analysis suggests that the model has predictive capabilities. Moreover, the study findings support the importance of perceived usefulness in technology-specific factors, facilitating conditions, and propagation in organizational-specific factors, government support, social influence, and external pressure in environmental-specific factors, as well as privacy concerns, self-efficacy, satisfaction, and domain-specific knowledge in end-user-specific factors. Four hypotheses related to perceived ease of use, service collaboration, habit, and innovativeness were rejected. Notably, the study identifies propagation as the most significant predictor in the ANN analysis. The conclusions of this study can be beneficial for university administrators, manufacturers, and policymakers in understanding the essential components of smart campuses to enhance the adoption and maximize the effectiveness of smart solutions.

**INDEX TERMS** Smart campus, IoT, adoption factors, smart education, higher education, SEM, ANN.

# I. INTRODUCTION

In recent years, the rapid progress of Internet of Things (IoT) technologies has opened up new possibilities for

The associate editor coordinating the review of this manuscript and approving it for publication was James Harland.

transformative applications across multiple domains [1], [2], including the realm of higher education. Particularly, the notion of a "Smart Campus" appeared as a promising paradigm, harnessing IoT capabilities to enhance campus operations, improve resource management, and provide innovative services to students and faculty [3]. An IoT-based

smart campus is described as the integration of IoT technologies and solutions in various aspects of a campus environment, such as education, administration, facilities management, and student services [3], [4], [5]. It involves the use of interconnected devices, sensors, and data analytics to improve operational efficiency, enhance safety and security, optimize resource utilization, and provide a seamless and personalized experience for students, faculty, and staff. The smart campus offers various benefits and functionalities services [3], [4] as summarised in Figure 1.

The IoT technologies can enhance the learning experience by enabling smart classrooms equipped with interactive displays, digital whiteboards, and connected devices [6]. Students can access educational resources, collaborate with peers, and receive personalized feedback. IoT-based tracking systems can also monitor attendance, analyze student engagement, and provide insights to improve teaching methodologies. Moreover, IoT devices and sensors can monitor the campus environment for potential risks, including surveillance cameras, access control systems, and fire detection sensors [3], [5], [7]. Specifically, devices like unmanned aerial vehicles (UAVs) have the potential to carry out surveillance and real-time monitoring, generating fresh information, as enlightened upon in the existing studies [8], [9], [10], [11], [12]. The real-time data collected from these devices can be analyzed to detect anomalies, raise alerts, and facilitate rapid response in case of emergencies in Campuses. Smart lighting systems can optimize energy consumption and enhance safety by automatically adjusting lighting levels based on occupancy and daylight.

In addition, the IoT-based solutions enable efficient management of campus infrastructure [3], which offers predictive maintenance to identify potential equipment failures and enable proactive maintenance, minimizing downtime [13]. IoT-based smart cards or wearables can serve as student identification, granting access to various campus facilities, such as libraries, labs, and dormitories [14]. Location-based services can provide real-time information about campus events, transportation options, and personalized notifications based on individual preferences [5], [15]. Furthermore, smart campus solutions can contribute to sustainable practices on a smart campus [3]. For example, smart grids and energy management systems optimize energy distribution and consumption, reducing carbon footprint. Environmental sensors can monitor air quality, temperature, and humidity, facilitating informed decisions for sustainable campus planning. Water management systems can monitor usage, detect leaks, and optimize irrigation in landscaping. Therefore, understanding the factors influencing Smart Campus adoption is crucial for educational institutions seeking to leverage these technologies effectively [16], [17], [18], [19].

To comprehensively investigate the factors influencing the IoT-based Smart Campus adoption intention, a robust analytical approach is required. This research paper presents multiple analysis methods, combining the power of Structural Equation Modelling (SEM) as well as Artificial Neural Networks (ANN) [20], [21], [22], [23], [24]. This integrated approach offers a more accurate insight into the complex dynamics underlying the adoption model and processes. In the first stage of the analysis, SEM is employed for the evaluation of the latent variables and observed variable relationships. SEM provides a statistical framework for testing theoretical models, enabling the assessment of the direct and indirect effects of various factors of IoT-based Smart Campus adoption. By capturing both the measurement as well as the structural models, the SEM offers a holistic view of the relationships among the constructs and enables the examination of hypothesized relationships.

Building upon the findings from the SEM stage, the second stage of analysis employs ANN to enhance the predictive power of the model. ANN leverages the principles of machine learning to evaluate and enhance the performance of predictive models by training neural network architectures. Through the ANN analysis, the importance of different predictors can be assessed, potentially uncovering hidden patterns and nonlinear relationships that may be missed by traditional statistical approaches. The integration of the two methods in this study contributes to a more robust and comprehensive investigation of the factors influencing the adoption of Smart Campus technologies. By combining the strengths of both approaches, we aim to provide valuable insights and practical implications for educational institutions embarking on Smart Campus initiatives. Understanding the drivers and barriers to adoption is essential for informed decision-making, resource allocation, and effective implementation strategies.

Hence, the central premise of this study revolves around exploring the significance of IoT-based smart campus adoption, a topic that has been gaining increasing attention in recent years due to rapid technological advancements and their implications for educational institutions. The novelty of this study comes from two key aspects; A) a holistic model for IoT-based Smart Campus adoption, and B) methodological design. This study believes that these elements collectively establish the novelty of the study and its potential to inform university administrators, manufacturers, policymakers, and researchers about the essential components of smart campuses.

#### A. HOLISTIC IOT-BASED SMART CAMPUS MODEL

The integration of factors from these four key themes results in a comprehensive and holistic framework that has not been extensively explored in the context of IoT-based smart campus adoption. While previous research might have focused on individual elements within these themes [3], [25], [26], [27], this study uniquely combines them through a systematic review process [17] and analytical hierarchical process [28] to provide a more complete understanding of the factors that collectively influence the adoption of smart campus solutions. This holistic approach reflects the intricate interplay between technology, organizational dynamics, environmental influences, and end-user perceptions empirically, thereby offering a richer and more nuanced perspective on



FIGURE 1. Summary of smart campus functionalities and benefits.

the adoption process. Notably, the integration of factors from diverse themes inherently brings together insights from multiple disciplines such as technology adoption, organizational behavior, environmental studies, and user experience. This interdisciplinary approach enriches the study's novelty by acknowledging the multifaceted nature of the topic. The incorporation of these different perspectives enables the study to contribute to a broader range of academic and practical domains, fostering a more holistic understanding of the challenges and opportunities presented by IoT-based smart campus adoption. Similarly, the incorporation of these four themes in the study's framework adds to its novelty by ensuring its applicability to various contexts. While the study focuses on universities in Iraq, the identified factors and their interrelationships can serve as a foundation for examining smart campus adoption in other regions and educational settings. This adaptability enhances the study's potential impact beyond its immediate context and strengthens its relevance to a wider audience.

# **B. METHODOLOGICAL DESIGN**

Considering the fact that the concept of smart campuses and the integration of IoT technologies have been explored to some extent [3], [5], [6], [29], [30], [31], [32], this study takes a distinct approach by focusing on a specific geographical context - universities in Iraq, with empirical data. This contextualization is crucial as it acknowledges the diversity in adoption challenges and opportunities that may not be directly transferable from other regions. As such, the study fills a gap by providing insights into the adoption of smart campus solutions in a region that might have unique sociocultural, economic, and technological factors.

Furthermore, the study's methodology adds to its novelty. The two-stage approach involving SEM followed by ANN analysis showcases a rigorous analytical process [33], [34], [35], [36], [37] that offers a comprehensive understanding of the factors influencing smart campus adoption. The ANN analysis, in particular, provides a predictive dimension to the study, enabling accurate forecasts and highlighting the potential of this approach for future research in similar domains. In addition, the empirical evidence collected from stakeholders affiliated with universities in Iraq, resulting in a dataset of 442 observations, further enhances the novelty of the study. The specific findings, including the identification of propagation [3] as a significant predictor, emphasize the importance of tailoring smart campus strategies to the local context. These elements collectively establish the novelty of the study and its potential to inform university administrators, manufacturers, policymakers, and researchers about the essential components of smart campuses for future consideration.

# C. ORGANISATION OF THE PAPER

In this study, the aim is to contribute significantly to the ever-growing body of knowledge on the adoption of IoT-based Smart Campus, offering insights that can drive successful implementation and utilization of these transformative technologies in the higher education landscape. Accordingly, the remaining part of the paper is structured as follows: Section II provides a review background of the relevant literature on IoT-based Smart Campus adoption as well discussion of the hypothesized model. Section III comprehensively describes the proposed research methodology, including data collection procedures, measurement instruments, and the analytical framework. Section IV presents the results and findings from the two-stage analysis, which covers data normality, measurement model, structural model, and ANN outcomes. Section V covers the result of the as well as implications for policy and practice for educational institutions. Finally, Section VI covers the conclusion of the study, which summarizes the key contributions and suggests avenues for future research.

# **II. LITERATURE OVERVIEW**

The majority of existing technology adoption studies have primarily revolved around the realm of IoT applications [38], [39], [40], [41], [42], [43] with less emphasis on IoT-based Smart Campus. Table 1 presents the current literature concerning the IoT smart campus. Accordingly, there's been a notable scarcity of studies focusing on the adoption context of Smart Campus. This perspective not only encompasses the incorporation of IoT technologies but also addresses various facets of campus life, from educational enhancements to effective infrastructure management. As a result, the proceeding sections provide a succinct overview of the research landscape. This underscores the crucial gap in the existing literature, namely the limited exploration of the broader smart campus adoption concept and hypothesis development.

# A. RESEARCH GAP AND MOTIVATIONS

The trends in smart campus research indicate a growing interest in leveraging IoT technologies, big data, and edge computing to enhance various aspects of campus life, from teaching and learning to infrastructure management. These trends demonstrate the potential for smart campuses to provide more efficient and user-centric services. While there are several noteworthy trends and areas of focus within the field. These trends provide valuable insights into the current state of research and the areas that require further exploration, especially highlighting the need for an integrated adoption model for IoT-based smart campuses, as presented in Table 1. For example, one of the focuses of the literature is proposing and experimenting with algorithms for smart campuses to improve features such as security, authentication, campus management, etc. [43], [44], [45], [46], [47], [48], [49], [50], [51], [52], [53], [54]. Other studies propose the criteria and indicators for successful smart campus initiatives [25], [55], [56], [57], [58], [59]. There is extensive literature on the theoretical or conceptual framework for smart campuses [56], [60], [61], [62], [63], and utilizing multi-decision criteria to investigate the factors for smart campuses [28], [64], and [65]. Specifically, several studies, e.g., [43],

[66], have explored the implementation of IoT in the educational sector. This approach aims to enhance traditional teaching methods with IoT technologies, improving student engagement and learning outcomes. The concept of a campus edge computing network, using elements such as street lighting as IoT communication nodes, has gained traction in recent research [44]. Similarly, some existing papers, e.g., [45], [64], emphasize the importance of robust data acquisition, management, and storage systems in smart campuses. Additionally, addressing semantic interoperability issues is a critical aspect of IoT-enabled smart campuses, as highlighted by Reference [63]. These issues relate to the compatibility and seamless communication of various IoT devices and applications within the campus ecosystem.

Moreover, the integration of IoT and big data analytics has been explored in creating small-scale testing environments for smart city technologies within a university campus [47], [60]. This approach focuses on optimizing resource management and supporting strategic decisions for improving campus services and sustainability. Several studies [55], [57], [61] focus on defining criteria and KPIs for evaluating smart campuses. These criteria help in assessing the success, effectiveness, and decision-making of smart campus initiatives. In addition, a human-centered approach to smart campus development is evident in studies evaluating user experiences and satisfaction [50]. This approach emphasizes the importance of meeting user expectations and requirements. The analysis of the prior research revealed that, while existing studies provide valuable insights into various aspects of smart campuses, there is a noticeable gap in the literature regarding an integrated adoption model for IoT-based smart campuses. The existing research focuses mostly on implementation, but there is a need for an adoption model for the successful deployment of smart campuses. Therefore, an integrated adoption model for IoT-based smart campuses serves as a valuable resource for universities planning to embark on the journey toward a smarter campus, and proposing such a model is a novel and significant contribution to the literature. The model would provide a roadmap for the strategic deployment of IoT technologies, considering all relevant components and their interplay. By addressing this research gap, this study contributes to the development of a comprehensive framework that guides universities in creating truly smart campuses that benefit both students and faculty.

#### **B. HYPOTHESIS DEVELOPMENT**

The literature analysis conducted in earlier studies reveals that studies on technology adoption can be categorized into four primary classifications [17], [28], [70], [71], [72], [73], [74]. These classifications, as previously discussed [17], [28], encompass four broad themes: technology-specific factors (TSF), organizational-specific factors (OSF), environmental-specific factors (ESF), and end-user-specific factors (USF) are the key elements considered. In adoption

# IEEE Access

# TABLE 1. Summary of smart campus existing studies.

Ref	Year	Paper summary	Focus/ con- tribution	Paper type/ method
[66]	2017	This paper introduces a model of the IoT and discusses the implementation of various IoT applications in a university campus, including IoT Flipped Classroom, IoT Entrance System, student feedback, IoT Orangery, and IoT heating system. The paper also includes feedback from students to assess the impact of implementing IoT-based Flipped Classroom compared to traditional educational approaches.	IoT-based Flipped Classroom	Implementation and student feedback
[44]	2018	This paper introduces a campus edge computing network that employs street lighting as IoT communication nodes, integrating various comput- ing services. Neural network learning algorithms analyze networks and compute resource requirements for efficient resource allocation and load balancing.	Campus edge computing network	Implementation and experiment
[45]	2018	This paper focuses on the design of an underlying data acquisition network comprising rules, drivers, and online equipment management to enable the collection of various data types. The Hadoop distributed system is employed for data storage and management, and data models encapsulate services, ultimately providing a safe and accurate service to smart campus users.	Data acquisition network	Implementation and experiment
[63]	2019	This paper focuses on addressing semantic interoperability issues in IoT- enabled smart campuses by identifying requirements for a semantic model to facilitate seamless and heterogeneous communication.	Semantic interoper- ability	Conceptual- based ontology
[46]	2019	This paper outlines the University of Málaga's long-term dedication to Smart Campus development, covering infrastructure, management, research support, and learning initiatives.	IoT and telecom- munication architecture	Implementation, case study
[47]	2019	The study explores the integration of IoT and big data in a traditional university campus to create a small-scale testing environment for smart city technologies. By collecting data through IoT devices and using big data analytics, the university campus serves as a prototype for optimizing resource management and sustainability in urban areas.	integration of IoT and big data	Prototype implemen- tation
[61]	2020	This study proposes a strategic map for intelligent university development, which underlines the pivotal role of technology in shaping successful smart campus projects	Strategic maps	Qualitative, theoretical, or conceptual
[25]	2020	The study combines qualitative and quantitative methods to analyze percep- tions and gather insights from stakeholders. The research results define and evaluate the criteria for a Smart Campus but also offer recommendations for the implementation of Smart Campuses in higher education settings.	Smart cam- pus criteria	Qualitative and quantitative analysis
[55]	2020	This research addresses the need for a comprehensive set of Key Perfor- mance Indicators (KPIs) for smart campuses and smart microgrids, which creates a list of 74 relevant and measurable KPIs that are categorized into 15 service areas. The objective is to provide a mechanism for campus management to assess and enhance the smartness of their university campus and microgrid.	Key Per- formance Indicators (KPIs)	Qualitative analysis
[43]	2020	This paper discusses a smart campus system in Chinese colleges that use campus big data, including various features related to student consumption, location preferences, and activities, to classify students and determine sub- sidies. It tackles data imbalance issues with different processing methods and employs classification algorithms for improving student management efficiency and staff workload reduction.	Smart cam- pus system	Implementation and experiment
[56]	2020	This study proposes a strategic framework for Smart Campus components. The framework centers on IoT and cloud computing, supported by eight main criteria and 25 sub-applications	Strategic framework	Theoretical or conceptual

# TABLE 1. (Continued.) Summary of smart campus existing studies.

[48]	2021	This paper addresses disputes in smart city application acceptance by proposing SpecTalk, which demonstrates it feasibility and effectiveness in conforming IoT implementations to TAICS standards for smart campus applications.	SpecTalk	Implementation/ experiment
[49]	2021	This article discusses integrating IoT sensors to monitor indoor air quality and adjust Heating, ventilation, and air conditioning (HVAC) systems in educational buildings, aiming to maintain well-being and enhance learning performance through training an artificial neural network.	HVAC sys- tems	Implementation and experiment
[50]	2021	This paper discusses the evaluation of user experiences using the MeCUE technique on the UMS HopIn! bus tracking app, part of the smart campus implementation at Universiti Malaysia Sabah. User evaluations reveal that the application met expectations, with most requirements fulfilled and a positive UX rating.	Bus tracking app	Implementation and experiment, case study
[60]	2021	This research explores existing IoT applications' capabilities and identi- fies the gaps in integrating IoT data into organizational decision-making processes. Through four case studies, it analyzes the decision-making processes and outlines the requirements for integrating IoT data into campus management for informed strategic decisions.	Requirements for integrating IoT data	Case study
[67]	2021	This study uses a systems thinking approach to identify SWOT factors affecting Smart Campus transitions. Data from a stakeholder workshop at the University of Technology were analyzed to create a causal loop diagram (CLD) highlighting the relationships between these factors.	Factors affecting Smart Campus transitions	Systems thinking approach
[58]	2021	This study examines stakeholders' perspectives on digital technologies and platforms for transitioning to smart campuses, focusing on a South African University of Technology. Data collected from staff and student representatives indicates a low level of awareness and utilization of digital technologies and platforms.	Perspectives on tran- sitioning to smart campuses	Survey
[51]	2022	This paper presents an improved algorithm for wireless sensor networks in smart campuses.	Security al- gorithm	Experiment
[57]	2022	This paper explores the role of financial information in creating intelligent campuses within the context of smart cities. It identifies key principles and tactical approaches, highlighting IoT and cloud technology as essential infrastructure.	Framework with eight main criteria and 25 sub- applications	Survey
[52]	2022	This paper introduces a Smart Campus system using spatiotemporal authen- tication fingerprints from GPS and CCTV to detect unauthorized access, making it suitable for Smart Cities 2.0 universities.	Security system	Implementation and experiment
[28]	2022	This study prioritizes the adoption factors for IoT-based smart campuses using an analytical hierarchical process (AHP).	Prioritisation of adoption factors	Investigation using AHP
[64]	2022	This study defines criteria for a 'Smart Campus' using AHP analysis with input from students, faculty, administrative staff, and IT support personnel at a UAE-based institute. The study ultimately developed a decision support tool based on the Utility function model to assist decision-makers in choosing the optimal solution for smart campus transformation.	Decision support tool	Investigation using AHP
[59]	2022	This research conducted an importance-performance analysis (IPA) at a Brazilian university, confirming the importance of eight dimensions for evaluating a smart campus and providing priority insights for academic managers.	Dimensions smart campus evaluation	IPA analysis
[53]	2023	This study introduces a mathematical decision-making tool, based on the evidential reasoning (ER) approach and implemented in Python, to help universities prioritize smart campus solutions.	Mathematical decision- making tool	Implementation and experiment, survey and validation

#### TABLE 1. (Continued.) Summary of smart campus existing studies.

[65]	2023	The study introduces a Smart Availability Scale (SAS) for evaluating	Smart	Multi-
		The study provides guidance for stakeholders transitioning from traditional to smart universities.	Scale	decision- making
[54]	2023	The paper discusses Universiti Utara Malaysia's adoption of IoT for cre- ating a smart campus, with a particular focus on an IoT-based campus bus tracking system using LoRa technology.	Tracking system	Implementation and experiment, case study
[68]	2023	The study explores how technologies can be applied to smart campuses. It presents findings related to IoT-based drone systems for patrolling, cloud servers for energy monitoring, AI for campus placement predictions, and more.	Technologies application	Qualitative
[69]	2024	The paper proposes a feedback-driven concept drift detection and adapta- tion methodology for real-world data streams environmental sensors at the University of Oulu Smart Campus.	Architecture for drift detection system	Implementation and experiment, case study

studies, it is crucial to consider various aspects related to Smart Campus, such as the technology, organization, environment, and end users. These factors collectively contribute to the successful implementation and utilization of technological solutions. For example, technology-specific factors refer to the specific features and capabilities of the technology that are relevant to the users, which involves understanding the functionalities, ease of use, compatibility with existing systems, security measures, and any unique attributes that make the technology appealing and beneficial [74], [75]. Organizational-specific factors focus on the characteristics and resources of the organization, particularly in the context of higher education institutions, such as the organization's size, degree of centralization, formalization, human resources, managerial structure, availability of slack resources, and the level of employee linkages [74]. The environmental factors encompass the external factors that surround the organization, including the structure and size of institutions, competitive landscape, regulatory environment, and macroeconomic background [73], [74]. Finally, end-user factors focus on the personal features as well as characteristics of the users themselves, such as their knowledge, skills, attitudes, perceptions, and preferences towards technology play a significant role in determining their acceptance and effective use of the technology [73]. By considering these factors, higher institutions can develop strategies that address the diverse aspects influencing the adoption and effective use of technology. Hence, as presented in Figure 2, the hypothesized model for the adoption of IoT-based smart campuses based on these factors is conceptualized to illustrate the impact of TSF, OSF, ESF, and USF on the behavioral intention of IoT-based Smart Campus adoption (BISC).

### 1) TECHNOLOGY SPECIFIC FACTORS

The factors specific to technology encompass the perceived ease of use (PE) and perceived usefulness (PU) as defined in the Technology Acceptance Model (TAM) [76]. PE refers \_\_\_\_\_

**IEEE**Access

to an individual's subjective evaluation of the effort required to use a particular technology or system, while behavioral intention represents an individual's inclination to engage in a specific behavior [76], [77], [78], [79]. Understanding the impact of PE on behavioral intention is critical for predicting and promoting the adoption of IoT-based Smart Campus initiatives. Previous research conducted in various technological contexts has explored the relationship between PE and behavioral intention, providing valuable insights for the present study. The TAM sheds light on this relationship, asserting that PE significantly influences behavioral intention, suggesting that individuals are more likely to adopt a technology if they perceive it as easy to use. Numerous studies investigating the adoption of IoT-based technologies have applied TAM and consistently supported the influence of PE on behavioral intention [80], [81], [82], [83]. Additionally, the Unified Theory of Acceptance and Use of Technology (UTAUT) [84], integrates influential theories including TAM, and emphasizes the significance of PE in shaping behavioral intention. According to UTAUT, perceiving technology as easy to use leads to more positive intentions to adopt it. Research applying UTAUT in the context of IoT-based technologies consistently reveals a positive association between PE and behavioral intention [81], [82], [83], [85], [86]. Therefore, hypothesis 1 (H1) is suggested as follows:

• H1: There is a significant relationship between perceived ease of use and BISC.

Secondly, perceived usefulness (PU) refers to an individual's subjective evaluation of how using a specific technology or system will surely enhance their performance or productivity, while behavioral intention represents an individual's inclination to engage in a particular behavior [76], [78], [79], [87], [88], [89], [90]. Understanding the influence of PU on behavioral intention is crucial for predicting and promoting the adoption of Smart Campus initiatives. Similarly, the TAM offers a theoretical framework to explore the relationship



FIGURE 2. Conceptual model and hypotheses.

between PU and behavioral intention. TAM highlight that individuals or peoples are more likely to adopt a technology if they perceive the technology to be useful. Numerous studies have applied TAM to investigate the adoption of IoT-based technologies and consistently support a positive relationship between PU and behavioral intention [87], [88], [89], [90], [91]. Apart from TAM, other theoretical models and frameworks also contribute to understanding the relationship between PU and behavioral intention. Notably, the UTAUT suggests that when individuals perceive technology as useful, they are more likely to develop positive intentions to adopt it. Several studies applying UTAUT in the context of IoT-based technologies have also found a positive relationship between PU and behavioral intention [87], [88], [89], [90]. This study specifically focuses on the adoption of IoT-based Smart Campus initiatives, highlighting the significance of PU in driving behavioral intention. Therefore, this study suggests hypothesis 2 (H2) as follows:

• H2: There is a significant relationship between perceived usefulness and BISC.

#### ORGANIZATIONAL SPECIFIC FACTORS

The organization-specific factors comprise facilitating conditions (FC), service collaboration (SC), and propagation (PG). Firstly, FC refers to the resources, support, and infrastructure available to individuals that facilitate the adoption and use of a technology or system [78], [84], [86], [92]. Understanding how FC influences behavioral intention is essential for promoting the successful implementation of IoT-based Smart Campus initiatives. The UTAUT also sheds light on the facilitating condition-behavioral intention relationship. UTAUT suggests that FC are critical determinants of individuals' intentions to use technology. When individuals perceive that the necessary conditions for using technology are in place, their behavioral intention to adopt and use the technology increases. Studies applying UTAUT in the context of IoT-based technologies environments have similarly found a positive association between FC and behavioral intention [78], [86], [92]. In the specific context of IoT-based Smart Campus adoption, research has emphasized the importance of FC in shaping behavioral intention. Hence, this study of IoT-based technologies and Smart Campus environments can be impacted by FC. Thus, this study posits hypothesis 3 (H3) as follows:

• H3: There is a significant relationship between facilitating conditions and BISC.

Secondly, service collaboration refers to the cooperation and interaction among different stakeholders, such as students, faculty, staff, and administrators, in utilizing and contributing to IoT-based Smart Campus services and applications [81]. Understanding how service collaboration influences behavioral intention is crucial for promoting effective implementation and utilization of IoT-based solutions for Smart Campus. Research specific to IoT-smart devices adoption highlights the importance of service collaboration in shaping behavioral intention [81]. In this study, increased collaboration and partnership between various entities (e.g. government organizations, non-governmental organizations, etc.) with the aim of sharing responsibility, authority, or support in governmental processes as well as actions, has a positive impact on smart adoption, including the adoption of smart campuses. Therefore, hypothesis 4 (H4) is proposed as follows:

• **H4:** There is a significant relationship between service collaboration and BISC.

Thirdly, propagation refers to the spread and dissemination of information, knowledge, and awareness about IoT-based Smart Campus initiatives among relevant stakeholders, such as students, faculty, staff, and administrators [3]. Understanding how propagation influences behavioral intention is crucial for effectively promoting and encouraging the adoption of IoT-based Smart Campus solutions. Existing studies specific to IoT-based Smart Campus adoption highlights the importance of propagation. For example, the emphasis on propagation especially to replicate smart campus was highlighted strongly in the prior study [3]. In this study, propagation has been suggested as one of the factors influencing the adoption of IoT-based Smart Campus technologies. Hence, hypothesis 5 (H5) is posited as follows:

• H5: There is a significant relationship between propagation and BISC.

#### 3) ENVIRONMENTAL SPECIFIC FACTORS

Environmental-specific factors encompass three key constructs, which include government support (GS), social influence (SI), and external pressure (EP). Government support refers to the initiatives, policies, funding, and resources provided by government entities to promote and facilitate the technologies and services of Smart Campus adoption [73], [87], [93]. Understanding how government supports influences behavioral intention is crucial for promoting the successful implementation and widespread adoption of IoT-based Smart Campus solutions. Accordingly, the presence of government support plays a significant role in shaping individuals' behavioral intention by providing a conducive environment and resources for IoT-based Smart Campus adoption [93]. Several studies have highlighted the impact of government support on behavioral intention [73], [87], [93]. Government support can take various forms, including policy initiatives, funding schemes, infrastructure development, and regulatory frameworks. These measures create an enabling environment and provide the necessary resources and incentives for stakeholders to adopt IoT-based Smart Campus solutions. Hence, hypothesis 6 (H6) is suggested as follows:

• **H6:** There is a significant relationship between government support and BISC.

Secondly, social influence refers to the impact of social interactions, norms, and the opinions of others on an individual's decision-making and behavior [86], [94], [95]. Therefore, understanding how social influence affects behavioral intention is crucial for promoting and facilitating the successful adoption of IoT-based Smart Campus solutions. The prominent Theory of Planned Behavior (TPB) [79], emphasizes the role of subjective norms in shaping behavioral intention. Subjective norms represent an individual's perception of the social pressures and expectations regarding a specific behavior. In the context of IoT-based Smart Campus adoption, subjective norms capture the influence of peers, colleagues, and the institutional community on an individual's behavioral intention. Studies applying TPB in the context of technology adoption have consistently found that subjective norms significantly influence behavioral intention [83], [85], [86], [92], [94], [95]. Therefore, this study highlights the significance of SI as a driver of behavioral intention among stakeholders in IoT-based smart campuses. As a result, this study proposed hypothesis 7 (H7) as follows:

• **H7:** There is a significant relationship between social influence and BISC.

In addition, external pressure refers to the influence exerted on individuals or organizations by external factors, such as regulatory bodies, industry standards, market demands, and stakeholder expectations [88], [93]. Hence, understanding how external pressure influences behavioral intention is crucial for effectively promoting and facilitating the adoption of IoT-based Smart Campus solutions. In the context of IoT-based Smart Campus adoption, external pressure captures the influence of external factors and stakeholders on individuals' behavioral intentions. Studies applied in technology adoption contexts have found that external pressure significantly influences behavioral intention [72], [87], [88], [93]. Therefore, the literature supports a relationship between external pressure and behavioral intention in the context of IoT-based Smart Campus adoption. As a result, hypothesis 8 (H8) is proposed as follows:

• **H8:** There is a significant relationship between external pressure and BISC.

#### 4) END-USER SPECIFIC FACTORS

The end-user factors (ESF) encompass the personal features or characteristics of the users. These factors cover privacy concerns (PC), satisfaction (SF), self-efficacy (SE), domainspecific knowledge (DK), habit (HB), and innovativeness (IN). Firstly, privacy concern refers to individuals' worries or apprehensions regarding the collection, disclosure, and use of their personal information in the context of IoT-based Smart Campus technologies and services [89], [92], [96]. Therefore, understanding how privacy concern influences behavioral intention is crucial for addressing privacy-related challenges and promoting the successful adoption of IoT-based Smart Campus solutions. Privacy concerns have emerged as a critical factor influencing individuals' behavioral intention to adopt IoT-based Smart Campus initiatives [89]. As the collection and utilization of personal data become more prevalent in educational settings, stakeholders become increasingly concerned about the potential risks associated with privacy breaches and unauthorized access to their information. Several studies have examined the impact of privacy concerns on behavioral intention in the context of IoT Smart technologies [71], [72], [73], [82], [89], [90], [92], [96], [97], [98], [99], [100]. The educational institutions and policymakers play a crucial role in addressing privacy concerns and shaping individuals' behavioral intention. The privacy concerns can be addressed through various strategies and measures to enhance individuals' trust and confidence in IoT-based Smart Campus initiatives. Therefore, this study proposed hypothesis 9 (H9) as follows:

• **H9:** There is a significant relationship between privacy concerns and BISC.

In the context of satisfaction and behavioral intention of smart campus relationships. Satisfaction refers to an individual's subjective evaluation of their experience, benefits, and overall fulfillment derived from using IoT-based Smart Campus technologies and services [101], [102]. Existing research has highlighted the importance of user satisfaction as a key element for achieving success in the field of information technology (IT) [103], [104], [105]. Within technology adoption studies, SF consistently emerges as a prominent factor due to its substantial influence on adoption behavior, as demonstrated in the literatures [106], [107], and [108]. Hence, understanding the influence of satisfaction on behavioral intention is crucial for ensuring the long-term success and sustainability of IoT-based Smart Campus solutions. Numerous studies have examined the impact of SF on behavioral intention in the context of technology adoption [101], [102]. Existing research consistently demonstrates a relationship between SF and behavioral intention, suggesting that individuals who are satisfied in terms of their previous and current experience are more likely to continue using and adopting the technology. Therefore, this study proposed hypothesis 10 (H10) as follows:

• **H10:** There is a significant relationship between satisfaction and BISC.

Thirdly, self-efficacy refers to individuals' belief in their own ability to successfully perform a specific task or behavior [73], [102], in this case, the adoption and usage of IoT-based Smart Campus technologies and services. Thus, understanding the influence of self-efficacy on behavioral intention is crucial for promoting individuals' confidence and motivation to adopt IoT-based Smart Campus solutions. Nevertheless, the literature consistently demonstrates a relationship between self-efficacy and behavioral intention in technology adoption contexts services [73], [95], [102], [109]. When individuals possess a high level of self-efficacy, they believe in their ability to overcome challenges, learn new technologies, and successfully adopt and utilize them. This confidence influences their intention to engage in the behavior. Therefore, individuals with higher self-efficacy are more likely to have stronger intentions to adopt and utilize IoT-based Smart Campus technologies and services. As a result, this study deposited hypothesis 11 (H11) as follows:

• **H11:** There is a significant relationship between self-efficacy and BISC.

The domain-specific knowledge refers to individuals' understanding and expertise in a particular subject or domain, in this case, the knowledge related to IoT-based Smart Campus technologies and their application in educational settings [109], [110]. Understanding the influence of domain-specific knowledge on behavioral intention is crucial for promoting individuals' competence and confidence in adopting and using IoT-based Smart Campus solutions. Various studies have explored the impact of domain-specific knowledge on behavioral intention in the context of technology adoption [109], [110]. The studies suggest a relationship between domain-specific knowledge and behavioral intention. This indicates that individuals with a higher level of insight as well as more knowledge in the IoT-based Smart Campus technologies are very likely to have a stronger intention to adopt and utilize these technologies. Therefore, hypothesis 12 (12) is suggested as follows:

• H12: There is a significant relationship between domain-specific knowledge and BISC.

Furthermore, habit refers to the automatic and repetitive behaviors that individuals develop over time, often driven by their previous experiences and established routines [86], [92]. Hence, understanding the influence of habit on behavioral intention is essential for identifying the factors that shape individuals' adoption patterns and promoting the sustained use of IoT-based Smart Campus technologies. Research examining the impact of habit on behavioral intention in technology adoption contexts [86], [92], has provided valuable insights into the relationship between habit and behavioral intention. The studies consistently demonstrate a relationship between habit and behavioral intention [86], [92]. This indicates that individuals with a strong habit of using technology are more likely to have a higher intention to adopt and utilize IoT-based Smart Campus solutions. Therefore, this study suggests that the extent to which users perform a particular behavior automatically or instinctively as a result of experience influences their behavioral intention to adopt IoT-based smart campus technologies. As a result, this study proposed hypothesis 13 (H13) as follows:

• **H13:** There is a significant relationship between habit and BISC.

Finally, innovativeness refers to individuals' propensity to adopt and embrace new technologies and innovations [110], [111], [112]. Similarly, understanding the influence of innovativeness on behavioral intention is crucial for identifying the factors that shape individuals' readiness to adopt and utilize IoT-based Smart Campus technologies. Existing research that investigated the impact of innovativeness on behavioral intention in technology adoption contexts, has provided valuable insights to suggest a relationship between innovativeness and behavioral intention [73], [90], [95], [110], [111], [112]. These studies suggest that individuals with a higher degree of innovativeness are more likely to have a stronger intention to adopt and utilize IoT smart technologies. Therefore, this study highlights that the willingness to accept IoT-based smart technology or service can influence users' behavioral intentions. Thus, this study posits hypothesis 14 (H14) as follows:

• **H14:** There is a significant relationship between innovativeness and BISC.

#### **III. METHOD**

This study proposes and analyzes a model that focuses on the adoption of IoT-based smart campuses. To achieve this objective, the study employs confirmatory factor analysis through the use of PLS-SEM, which stands for partial least squares structural equation modeling. In order to lay the groundwork for this research, a systematic literature review (SLR) of existing literature on technology adoption was conducted, focusing on identifying the influencing factors of IoT smart devices. The literature review conducted by Ref erence [17] provides a comprehensive list of influencing factors related to smart campus applications and implementations. To further evaluate the identified influencing factors and to facilitate the selection of the factors objectively, the researchers utilized an Analytical Hierarchy Process (AHP) [28], which is a widely recognized multi-criteria decision-making technique [70], [113]. By employing AHP, the study systematically evaluated and prioritized various factors involved in the selection process. This allowed for a structured and rigorous analysis, enabling the selection of the most suitable options based on their relative importance and performance according to the defined criteria [70]. This process involved classifying and ranking 25 technology adoption factors according to the influencing factors identified in Sneesl [28], a conceptual model consisting of 14 factors was developed to examine the proposed model. Furthermore, the model was then analyzed using PLS-SEM. It is worth mentioning that previous studies [24], [90], [95], [111], [114] have also utilized this approach. The proposed research method consists of three phases, as depicted in Figure 3, which are discussed in the proceeding subsections. The current study provides a further discussion of the chosen method.

# A. PHASE 1: LITERATURE REVIEW THROUGH SLR

The literature review was conducted following the guidelines of SLR as reported in previous SLR studies [115], [116], [117]. The SLR process was divided into four stages, as depicted in Figure 3 (see Sneesl [17] for more detail). Nevertheless, in the first stage, a comprehensive search for relevant research articles was conducted. Keywords identified from the existing literature were used to generate a query for searching research articles. The second stage involved the screening of articles based on their titles and abstracts, which excluded articles that are not relevant from further reading. Following the title and abstract screening, the selected articles underwent a more detailed evaluation based on criteria for inclusion and exclusion. Accordingly, the inclusion criteria strictly comprised papers indexed in JCR or Scopus and articles that incorporated technology acceptance or adoption theories. After applying these criteria, a final 108 articles remained for full-text reading and analysis. Finally, 59 articles were read in full. Accordingly, the literature reading and data extraction identified and collated various technology adoption factors. This study made a significant finding by identifying a total of 112 factors, which were classified into four themes [17] (see Figure 2). However, to ensure accuracy and avoid redundancy, duplicate factors were identified and filtered, resulting in a refined set of 77 factors. Further refinement was achieved through a second round of duplicate filtering using thematic analysis, which ultimately yielded 52 factors.

#### B. PHASE 2: RANKING OF FACTORS THROUGH AHP

In order to delve deeper into the 52 factors identified in a previous study [17], the AHP technique was employed in Reference [28]. The purpose was to simplify the analysis and focus on the most crucial factors for the implementation of smart campuses. Initially, an objective analysis was conducted using the frequency technique, which helped narrow down the selection to 25 factors for the subsequent AHP analysis [28]. The AHP methodology involves breaking down the decision-making problem into three levels, namely the objective, criteria, and alternatives [113]. By applying the AHP in the aforementioned study [28], the study was able to identify and determine the most significant contributing factors for the successful adoption of smart campuses. The AHP technique played a crucial role in systematically evaluating and prioritizing the identified factors, providing valuable insights for decision-makers and stakeholders involved in implementing smart campuses. Hence, the application of the AHP technique proved instrumental in identifying the key factors that contribute to the successful implementation of smart campuses. Notably, the global ranking of the factors suggested that twelve factors from various categories led to the conceptualization of the IoT-based smart campus model proposed in this study.

#### C. PHASE 3: EMPIRICAL STUDY

In this stage, two preliminary activities were performed before the data collection for the empirical study. These activities are briefly discussed which comprise instrumentation such as expert evaluation and pilot test. Moreover, the PLS-SEM analysis and artificial neural network analysis (ANN) methods are discussed accordingly.

# 1) INSTRUMENTATION THROUGH EXPERT REVIEW

Based on the literature for technology adoption for IoT and smart devices [82], [83], [85]. The instrument for this



FIGURE 3. Proposed research methodology.

study is developed, adapted, or adopted in accordance with academic research practice. Then, an expert review document was created for early content validity from a group of expert panels in the information systems, smart campus, IoT, and smart technologies fields. This involves seeking expert opinions in the relevant field of study [118], [119]. By incorporating expert judgments and recommendations, the proposed model is further validated by five experts. 14 factors were derived from technology adoption literature and AHP, as factors affecting the adoption intention of Smart Campus. The expert evaluation form was designed according to the classification of the factors proposed previously [17], [28]. The expert evaluation form was designed in such a way that the responses provide a valuable report on the content, face, and construct validity of the proposed model. Accordingly, the expert review form indicates the rating of each factor based on a 4-point Likert scale. Moreover, the form is divided into key sections. PART A covered the participant information. PART B: covered the construct and face validity of the IoT-based Smart Campus adoption model as well as a text box for experts to provide comments. Finally, PART C covered the content validity for each construct as well as the items. Accordingly, the expert response was evaluated through the content validity index (CVI) with a modified kappa statistical coefficient. By following these content validity recommendations and employing appropriate statistical measures like the CVI and kappa statistics, the reliability, as well as validity of the research items, are ensured for subsequent analyses.

#### 2) PILOT TEST DESIGN

The pilot test, often considered a crucial component of a good research design, plays a significant role in evaluating the research instrument's effectiveness [120]. Its purpose, as a preliminary investigation, is to test the viability and significance of the research instrument before conducting a full-scale study [120], [121]. In this study, after the instrument passes the content validity test is then adopted for the pilot test. The pilot test questionnaire includes multiple-choice questions and sections. The first section focuses on participant demographics which covers years of experience in the university. Accordingly, the validity of the instrument is thoroughly investigated, which is done to determine the instrument's reliability in order to ascertain the effectiveness of the items associated to each construct. To evaluate the instrument's reliability of the variables as well as the correlation between the variables, Cronbach's alpha and regression models were employed. Hence, the direct effect was tested through percentile bootstrapping (5000 samples) [122] using SPSS 24 with 35 participants. The decision to employ these indexes and analysis methods was influenced by their widespread use in the literatures [123],

[124], and [125]. In particular, Cronbach's alpha was utilized to verify the item's reliability, which ranged from 0.7 to 0.9, an indication that the reliability of the research instrument is acceptable [126].

# 3) SURVEY PROCEDURE

Staff from all the public universities in Iraq were the target participants in this study. Detailed demographic information about the participants is recorded in Table 2. The survey questionnaire was translated into the Arabic language of the target participants. The Arabic version of the questionnaire can be accessed through an electronic medium, particularly Google Forms, to collect the data. It can be found at the following link: online survey for a smart campus. Nevertheless, the survey questionnaire is presented in the Appendix. Accordingly, the survey methodology played a pivotal role in data collection and analysis. This study meticulously designed the data collection process, targeting a representative sample of participants from all the universities in Iraq. The survey instrument, a structured questionnaire tailored to the study's objectives about a smart campus, was crafted with precision to ensure that it would be easily understood by the participants. The questionnaire structure and content were fine-tuned to ensure it gathered the necessary data. As mentioned earlier, the study conducted a pilot test to refine the survey instrument before its full deployment, which is an essential step in optimizing the survey's effectiveness. The data collection procedure, primarily administered through online platforms, incorporated ethical considerations such as informed consent and data privacy statements. Throughout the survey, ethical considerations remained paramount, ensuring participant anonymity and adherence to ethical guidelines and regulations. To maintain data quality and reliability, rigorous data validation and quality control processes were in place. Statistical analysis was conducted to derive meaningful insights from the collected data, as discussed in the proceeding section. This transparent and well-documented survey procedure underpins the credibility and validity of our research findings.

# 4) DATA ANALYSIS PROCEDURE

After the data collected is completed, the analysis section covers the data screening procedures, measurement assessment, structural models, and ANN analysis. The data screening was specifically used to check the multicollinearity issues and common method bias (CMB). Additionally, the reflective measurement models deal with the models' reliability and validity, by analyzing the different types of tests via composite reliability (CR), convergent and discriminant validity, while the structural models test the hypotheses through path analysis, and model fit and prediction capacities through coefficients of determination ( $\mathbb{R}^2$ ), model fit indices, and F-test statistics ( $\mathbb{F}^2$ ).

Choosing an appropriate statistical model to analyze survey research has remained a challenge for researchers.

TABLE 2.	Demographic	distribution	of the	respondents.
----------	-------------	--------------	--------	--------------

Gender distribution	Male	266 (60.2%)
	Female	176 (39.8%)
Age Distribution	Less than 18	0 (0%)
	18 to 24 years	14 (3.2%)
	25-34 years	81 (18.3)
	35-44 years	155 (35.3%)
	45 years above	192 (43.4%)
Educational background	Doctoral degree	195 (44.1%)
	Master's degree	114 (25.8%)
	Bachelor's degree	97 (21.9%)
	Diploma	36 (7.9%)

Covariance-based structural equation modeling (CB-SEM) has traditionally been the dominant approach [127], but the PLS-SEM has gained popularity in recent years [127], [128]. In the current study, the proposed model was evaluated using SmartPLS 4.0 program, which employed the PLS-SEM method [127]. The model was computed using the bootstrapping method with 10000 samples [129]. PLS-SEM, a variance-based approach, accounts for all the variance and uses the sum of the variance to estimate model parameters. This choice is guided by the literature recommendation, which is suitable for testing the theoretical framework and predicting outcomes [127].

Finally, to assess the predictive capacity of the model, an ANN was employed. ANN modeling tool simulate human-like neural systems and learn the behavior of the agent under investigation [34]. To enhance the performance of neural networks, their learning capabilities can be trained [33], [35]. For the ANN analysis, SPSS v24 was utilized, following previous studies [20], [21], [22], [23], [24]. Specifically, an ANN using a feed-forward back-propagation algorithm was employed to estimate the relative significance of the independent variables against the dependent variable through the Multilayer Perceptron (MLP) method. A tenfold cross-validation procedure was performed on the dataset, resulting in 10 ANN models. Accordingly, the training phase utilized 70% of the data, while the remaining 30% was used to assess the projected fitness of the trained network, thereby minimizing the risk of overfitting.

# **IV. RESULT**

The structural assessment model plays a crucial role in analyzing the relationships and dynamics among variables within a research study [130]. Hence, the structural model allows this study to assess and predict the outcome variable. It examines the causal relationships to determine the strength and significance of the variables [131]. The structural assessment model aids in understanding how different factors interact and influence the outcome variable under investigation. Utilizing SEM, researchers can assess the overall fit of a given model and evaluate the significance and effect sizes of the hypothesized relationships [131]. These techniques provide insights into the complex relationships between variables, contributing to a deeper understanding of the underlying mechanisms and supporting evidence-based decision-making [132]. In this study, the assessment of the IoT-based smart campus model employed in the current study is presented, which includes the measurement model, path analysis, and the assessment of model fit and effect sizes. In addition, the assessment of ANN analysis was covered in this section. By applying these techniques, we aim to gain insight on the relationships between variables and their impact on IoT-based smart campus adoption.

### A. DATA NORMALITY

Before examining the proposed model, the collected data underwent several normality tests to assess whether any normality issues were present. Firstly, the data was subjected to Harman's single factor test, a commonly used method for evaluating common method bias (CMB) [133]. The test was conducted on all variables to determine the presence of CMB. The findings revealed that after rotating 79 factors, one factor accounted for approximately 26.75% of the variance. This value is below the established threshold of 50% ([133], [134], [135]. Consequently, it can be inferred that the data does not exhibit any concerns related to CMB. Secondly, the Bartlett Sphericity test was employed to examine whether the correlation matrix was formed by two or more independent sources. The obtained result yielded a significance level of 0.00, the data collected from universities staffs to determine their behavioral intention to use IoT-based smart campus, follows a normal distribution. Additionally, the Kaiser-Meyer-Olkin (KMO) measure of sample adequacy was also utilized to check the normality, resulting in a satisfactory value of 0.897 for the total correlation matrix (commonly referred to as "marvelous"). By considering these normality coefficients, it can be concluded that there are no normality issues present in this study, indicating that the model can be further analyzed.

# **B. MEASUREMENT MODEL ASSESSMENT**

After assessing the data for normality issues and obtaining satisfactory results, the next step involved subjecting the data to a reliability test to determine the convergent validity techniques. The result obtained for convergent validity analysis revealed that all item loading's exceeded 0.500, specifically ranging from 0.556 to 0.918, as depicted in Table 3 and Table 4. These results indicate that the items and their related constructs exhibit a significant degree of variance [127], [136]. However, it was observed that a few item loading values significantly impacted the reliability and validity coefficients, such as Cronbach's alpha, coefficients A and C, and average variance extracted (AVE). Consequently, these items were systematically eliminated one at a time, resulting in the removal of a total of 15 items (PU1, GS2, GS3, GS6, PC4, PC5, PC7, SF6, SF7, DK5, PG1, FC6, FC7, FC8, and BISC1). The reliability metrics for the latent variables of the model are summarized in Table 3. Moreover, Cronbach's alpha values ranged from 0.714 to 0.937, all exceeding the recommended threshold of 0.70, indicating a strong level

126008

of item reliability. Additionally, to demonstrate composite reliability, the CR-rho\_a coefficient should surpass 0.7, and in this study, it ranged from 0.747 to 0.971. Similarly, the composite reliability (CR-rho\_c) values were also higher than 0.700, ranging from 0.818 to 0.955. Furthermore, all AVE metrics were greater than 0.500, with values ranging from 0.534 to 0.841 [127], [136]. Therefore, the reliability test results, along with the convergent validity techniques, provide evidence that the measurement model employed in this study is robust and accurately captures the underlying constructs of the data.

The discriminant validity of the measures was assessed by comparing the indicator loadings of each construct to the others. As depicted in Table 4, the results indicate that all indicators loaded at or above the loading of the other constructs, demonstrate significant discriminant validity. Furthermore, the loadings of each indicator on its corresponding construct were significantly greater and more significant than the loadings of the indicators on the other variables in the sample, further supporting the presence of discriminant validity. In addition to the crossloading analysis, the Fornell-Larcker criterion was utilized to evaluate the discriminant validity of the measurement model, as presented in Table 5.

The Fornell and Larcker criterion is widely recommended for assessing the degree of shared variance among constructs [137], [138]. While it does not provide a definitive measure of differences between constructs, it is commonly employed due to its extensive use in the literature [131], [138], [139]. The Fornell and Larcker criterion was applied to evaluate discriminant validity. Accordingly, discriminant validity is evaluated by comparing the square roots of the AVE and the correlation coefficients between the constructs. Accordingly, the results obtained for the Fornell and Larcker criterion (see Table 5), indicates that the AVE values for all constructs exceed the threshold value, a criterion satisfied by this study findings [137]. Furthermore, the Fornell-Larcker index requires two criteria to establish discriminant validity. Firstly, the square root of each construct's AVE must be greater than its correlation with another construct. Secondly, each item must load disproportionately on the construct with which it is most closely correlated [136], [137]. Based on the Fornell-Larcker index results, all values are statistically significantly larger than the correlated constructs, indicating acceptable discriminant validity.

Moreover, the Heterotrait-Monotrait Ratio of Correlation (HTMT) was employed as a method to assess the correlation ratios between constructs [131]. HTMT is used to measure discriminant validity among variables [138]. In HTMT, a value of 0.90 or higher suggests a lack of discriminant validity, while a value below 0.85 is considered acceptable for demonstrating discriminant validity between two variables. Additionally, the HTMT method evaluates the average correlation of indicators among constructs in the model [138], [140]. It is worth noting that the HTMT confidence interval should not include a value of one (1), as a value close

# TABLE 3. Item loadings and reliability measures.

Construct	Items	Loadings	Cronbach's alpha	CR-rho_a	CR-rho_c	AVE
	BISC2	0.835	0.820	0.825	0.877	0.592
	BISC3	0.840				
Behavioural intention of SmartCampus	BISC4	0.821				
1	BISC5	0.758				
	BISC6	0.556				
	DK1	0.918	0.937	0.952	0.955	0.841
	DK2	0.895	0.957	0.952	0.955	0.011
Domain-specific knowledge	DK3	0.0934				
	DK4	0.934				
	ED1	0.921	0.761	0.761	0.862	0.676
Enternal macazina	EF I ED2	0.842	0.701	0.701	0.802	0.070
External pressure	EP2 ED2	0.850				
	EP3	0.793	0.021	0.042	0.000	0.504
	FCI	0.798	0.831	0.843	0.880	0.594
	FC2	0.730				
Facilitating condition	FC3	0.786				
	FC4	0.760				
	FC5	0.780				
	GS1	0.779	0.746	0.747	0.854	0.662
Government support	GS4	0.852				
	GS5	0.809				
	HB1	0.804	0.883	0.891	0.914	0.680
	HB2	0.851				
Habit	HB3	0.798				
	HB4	0.843				
	HB5	0.827				
	IN1	0.779	0 774	0.777	0.870	0.690
Innovativeness	IN2	0.862	01771	01777	0.070	0.070
inito vali venesis	IN3	0.849				
	PC1	0.831	0.714	0.759	0.818	0.534
	PC2	0.604	0.714	0.757	0.010	0.554
Privacy concern	DC2	0.652				
	PC6	0.052				
	PC0	0.809	0.914	0.910	0.970	0.574
	PEI	0.789	0.814	0.819	0.870	0.374
	PE2	0.780				
Ease of use	PE3	0.732				
	PE4	0.677				
	PE5	0.803				
	PG2	0.813	0.856	0.864	0.902	0.698
Propagation	PG3	0.872				
Topagation	PG4	0.858				
	PG5	0.798				
	PU2	0.791	0.826	0.826	0.885	0.657
	PU3	0.788				
Useruiness	PU4	0.840				
	PU5	0.822				
	SC1	0.702	0.790	0.800	0.864	0.615
	SC2	0.828			•	
Service collaboration	SC3	0.773				
	SC4	0.826				
	SC4 SE1	0.820	0.840	0.863	0.804	0.621
	SEI	0.000	0.047	0.003	0.074	0.031
Calf off agor	SEZ	0.884				
Sen-encacy	SE3	0.804				
	SE4	0.748				
	SES	0.627	0.020	0.071	0.045	0.550
	SF1	0.814	0.929	0.971	0.946	0.778
~	SF2	0.912				
Satisfaction	SF3	0.912				
	SF4	0.893				
	SF5	0.875				
	SI1	0.812	0.880	0.882	0.913	0.678
	SI2	0.839				
Social influence	SI3	0.868				
	SI4	0.861				
	SI5	0.731				
	~**					

to 1 indicates a lack of discriminant validity [140]. The correlations among the constructs based on the HTMT are presented in Table 6. The results indicate that all values

obtained for the constructs in this study were below the threshold of 0.85, suggesting that discriminant validity is achieved in the measurement models. Consequently, the

# TABLE 4. Crossloadings of measurement items.

	BISC	DK	EP	FC	GS	HB	IN	PC	PE	PG	PU	SC	SE	SF	SI
	0.835	0.331	0.454	0.346	0.457	0.441	0.456	0.602	0.355	0.498	0.355	0.390	0.397	0.073	0.264
	0.840	0.200	0.101	0.303	0.432	0.330	0.100	0.544	0.270	0.120	0.307	0.304	0.3/1	0.070	0.237
DICC	0.040	0.299	0.397	0.303	0.432	0.339	0.300	0.344	0.279	0.423	0.307	0.304	0.341	0.009	0.237
DISC	0.021	0.525	0.410	0.295	0.427	0.334	0.391	0.470	0.300	0.550	0.290	0.405	0.301	0.042	0.192
	0.738	0.239	0.554	0.270	0.575	0.275	0.302	0.435	0.210	0.454	0.234	0.415	0.304	0.097	0.207
	0.550	0.764	0.293	0.524	0.343	0.385	0.339	0.469	0.389	0.452	0.244	0.329	0.080	0.262	0.472
	0.467	0.918	0.278	0.4/1	0.285	0.294	0.435	0.443	0.429	0.424	0.174	0.300	0.686	0.340	0.503
DK	0.365	0.895	0.239	0.444	0.242	0.230	0.370	0.392	0.336	0.390	0.123	0.235	0.645	0.276	0.489
211	0.501	0.934	0.315	0.488	0.327	0.285	0.438	0.504	0.402	0.440	0.183	0.299	0.646	0.285	0.504
	0.537	0.921	0.314	0.520	0.365	0.362	0.433	0.514	0.434	0.468	0.239	0.323	0.659	0.364	0.555
	0.381	0.240	0.842	0.222	0.271	0.383	0.402	0.384	0.159	0.331	0.215	0.455	0.207	0.014	0.202
EP	0.401	0.218	0.830	0.183	0.314	0.386	0.480	0.395	0.123	0.380	0.207	0.443	0.223	0.074	0.224
	0.452	0.312	0.793	0.305	0.313	0.354	0.378	0.410	0.196	0.436	0.105	0.446	0.341	0.097	0.271
	0.314	0.353	0.213	0.798	0.229	0.329	0.259	0.476	0.380	0.379	0.121	0.296	0.387	0.228	0.402
	0.310	0.306	0.219	0.730	0.449	0.350	0.247	0.499	0.340	0.383	0.143	0.337	0.326	0.245	0.417
FC	0.444	0.549	0.251	0.786	0.362	0.358	0.374	0.548	0.507	0.440	0.248	0.324	0.627	0.281	0.460
	0.361	0.332	0.236	0.760	0.292	0.421	0.170	0.491	0.337	0.411	0.165	0.271	0.420	0.249	0.503
	0.293	0 447	0.194	0.780	0.331	0.310	0.121	0 549	0 334	0 357	0.148	0.222	0.512	0.224	0 559
	0.485	0.223	0.362	0.334	0.779	0.434	0.311	0.426	0.293	0.672	0.226	0.330	0.343	0.150	0.252
GS	0.414	0.253	0.320	0.358	0.852	0.191	0.311	0.445	0.270	0.459	0.220	0.396	0.277	0.150	0.232
00	0.414	0.255	0.520	0.363	0.052	0.392	0.302	0.443	0.270	0.457	0.201	0.300	0.277	0.134 0.237	0.349
	0.391	0.301	0.192	0.303	0.009	0.303	0.200	0.455	0.341	0.597	0.201 0.241	0.509	0.367	0.257	0.335
	0.404	0.201	0.440	0.377	0.408	0.004	0.440	0.301	0.312	0.347	0.241	0.528	0.307	0.100	0.340
UD	0.391	0.165	0.342	0.510	0.455	0.051	0.550	0.401	0.200	0.460	0.241	0.420	0.333	0.210	0.277
пв	0.310	0.237	0.328	0.308	0.249	0.798	0.279	0.388	0.240	0.382	0.125	0.349	0.279	0.238	0.289
	0.375	0.334	0.380	0.429	0.402	0.843	0.329	0.546	0.181	0.483	0.137	0.413	0.353	0.229	0.414
	0.383	0.324	0.365	0.422	0.335	0.827	0.289	0.489	0.235	0.408	0.155	0.366	0.331	0.185	0.357
	0.401	0.318	0.472	0.121	0.238	0.275	0.779	0.310	0.273	0.346	0.298	0.427	0.230	0.176	0.187
IN	0.429	0.414	0.393	0.276	0.323	0.383	0.862	0.390	0.397	0.415	0.315	0.369	0.304	0.221	0.312
	0.439	0.411	0.411	0.380	0.365	0.381	0.849	0.465	0.381	0.440	0.230	0.539	0.409	0.325	0.433
	0.614	0.421	0.434	0.467	0.348	0.538	0.443	0.831	0.329	0.525	0.256	0.419	0.485	0.190	0.423
PC	0.355	0.251	0.394	0.510	0.450	0.383	0.331	0.604	0.293	0.400	0.137	0.503	0.276	0.195	0.335
10	0.346	0.402	0.285	0.662	0.467	0.352	0.256	0.652	0.279	0.415	0.185	0.330	0.346	0.175	0.452
	0.567	0.412	0.312	0.428	0.387	0.455	0.326	0.809	0.230	0.402	0.182	0.342	0.411	0.262	0.445
	0.291	0.364	0.161	0.448	0.332	0.215	0.323	0.289	0.789	0.325	0.383	0.171	0.444	0.373	0.380
	0.263	0.338	0.069	0.399	0.251	0.158	0.184	0.224	0.780	0.309	0.486	0.037	0.469	0.289	0.408
PE	0.331	0.219	0.205	0.222	0.196	0.293	0.462	0.218	0.732	0.317	0.527	0.171	0.296	0.261	0.258
	0.289	0.306	0.163	0.458	0.342	0.189	0.304	0.351	0.677	0.332	0.213	0.181	0.300	0.246	0.384
	0.356	0.436	0.133	0.401	0.289	0.264	0.304	0.342	0.803	0.262	0.422	0.229	0.427	0.234	0.388
	0.435	0.332	0.286	0.498	0.579	0.445	0.330	0.471	0.388	0.813	0.201	0.320	0.438	0.221	0.341
	0.561	0.444	0.445	0.491	0.490	0.455	0.465	0.523	0.375	0.872	0.244	0.430	0.453	0.188	0.332
PG	0.552	0.419	0.368	0.376	0.605	0.481	0.363	0.472	0.381	0.858	0.254	0.414	0.515	0.163	0.338
	0.502	0.371	0.455	0.372	0.479	0.511	0.450	0.517	0.213	0.798	0.134	0.520	0.420	0.134	0.340
	0.316	0.160	0.100	0.116	0.273	0.159	0.150	0.229	0.213	0.237	0.101	0.177	0.120	0.026	0.0101
	0.310	0.100	0.190	0.245	0.273	0.159	0.260	0.269	0.460	0.207	0.791	0.238	0.335	0.020	0.101
PU	0.331	0.204	0.101	0.245	0.234	0.207	0.204	0.200	0.407	0.202	0.700	0.250	0.355	0.105	0.210
	0.298	0.108	0.157	0.100	0.178	0.193	0.313	0.171	0.427 0.442	0.195	0.040	0.100	0.195	0.001	0.040
	0.290	0.115	0.100	0.190	0.100	0.105	0.240	0.101	0.445	0.101	0.022	0.179	0.109	0.051	0.049
	0.520	0.241	0.300	0.414	0.554	0.444	0.344	0.445	0.145	0.404	0.177	0.702	0.244	0.254	0.235
SC	0.402	0.262	0.392	0.198	0.331	0.415	0.303	0.400	0.064	0.402	0.187	0.828	0.231	0.052	0.115
	0.363	0.235	0.450	0.317	0.237	0.398	0.477	0.394	0.206	0.401	0.1/1	0.773	0.318	0.119	0.212
	0.416	0.264	0.481	0.291	0.405	0.366	0.495	0.427	0.261	0.391	0.205	0.826	0.242	0.192	0.198
	0.456	0.619	0.254	0.541	0.262	0.289	0.297	0.452	0.428	0.457	0.190	0.229	0.880	0.214	0.540
	0.531	0.608	0.263	0.482	0.353	0.340	0.349	0.465	0.425	0.447	0.242	0.253	0.884	0.189	0.488
SE	0.432	0.603	0.247	0.562	0.255	0.276	0.368	0.411	0.416	0.413	0.177	0.314	0.804	0.270	0.493
	0.392	0.540	0.212	0.415	0.342	0.328	0.214	0.390	0.336	0.437	0.217	0.239	0.748	0.270	0.487
	0.400	0.470	0.290	0.392	0.314	0.419	0.271	0.382	0.408	0.424	0.261	0.272	0.627	0.212	0.429
	0.065	0.241	0.054	0.232	0.162	0.267	0.189	0.228	0.257	0.111	0.092	0.078	0.208	0.814	0.374
	0.147	0.359	0.058	0.294	0.177	0.179	0.261	0.248	0.329	0.198	0.055	0.167	0.323	0.912	0.450
SF	0.099	0.282	0.024	0.239	0.180	0.204	0.241	0.210	0.337	0.200	0.068	0.154	0.256	0.912	0.399
	0.109	0.279	0.083	0.315	0.207	0.254	0.267	0.280	0.337	0.192	0.137	0.175	0.217	0.893	0.431
	0.102	0.342	0.123	0.324	0.241	0.229	0.314	0.270	0.344	0.195	0.082	0.214	0.226	0.875	0.433
	0.005	0.504	0 299	0.531	0.337	0.434	0.338	0.507	0.434	0.395	0.099	0.269	0.545	0.401	0.812
	0.335	0.504	0.2//	0.001											
	0.335 0.280	0.471	0.145	0.471	0.261	0.282	0.298	0.410	0.435	0.283	0.152	0.103	0.475	0.378	0.839
SI	0.335 0.280 0.318	0.471 0.502	0.145	0.471 0.532	0.261 0.356	0.282 0.306	$0.298 \\ 0.275$	$0.410 \\ 0.483$	0.435 0.369	0.283 0.315	$0.152 \\ 0.060$	0.103 0.225	$0.475 \\ 0.499$	$0.378 \\ 0.387$	0.839 0.868
SI	0.335 0.280 0.318 0.280	0.304 0.471 0.502 0.482	0.145 0.226 0.232	0.471 0.532 0.514	0.261 0.356 0.307	0.282 0.306 0.359	0.298 0.275 0.334	$0.410 \\ 0.483 \\ 0.443$	$0.435 \\ 0.369 \\ 0.450$	0.283 0.315 0.364	$0.152 \\ 0.060 \\ 0.117$	0.103 0.225 0.146	$0.475 \\ 0.499 \\ 0.542$	0.378 0.387 0.447	0.839 0.868 0.861

#### TABLE 5. Fornell-Larcker criterion matrix.

	BISC	DK	EP	FC	GS	HB	IN	PC	PE	PG	PU	SC	SE	SF	SI
BISC	0.770														
DK	0.518	0.917													
EP	0.504	0.316	0.822												
FC	0.458	0.527	0.292	0.771											
GS	0.535	0.338	0.366	0.432	0.813										
HB	0.474	0.325	0.456	0.462	0.469	0.825									
IN	0.510	0.460	0.510	0.317	0.374	0.419	0.831								
PC	0.670	0.511	0.484	0.667	0.536	0.601	0.470	0.730							
PE	0.409	0.441	0.196	0.504	0.370	0.302	0.424	0.378	0.758						
PG	0.618	0.473	0.469	0.517	0.641	0.566	0.484	0.593	0.405	0.836					
PU	0.384	0.202	0.210	0.222	0.271	0.224	0.337	0.265	0.540	0.251	0.811				
SC	0.481	0.320	0.546	0.379	0.425	0.512	0.537	0.528	0.216	0.507	0.236	0.784			
SE	0.562	0.718	0.319	0.604	0.384	0.413	0.381	0.531	0.509	0.548	0.273	0.328	0.794		
SF	0.125	0.348	0.077	0.322	0.219	0.248	0.292	0.281	0.367	0.209	0.095	0.185	0.287	0.882	
SI	0.373	0.561	0.286	0.606	0.381	0.410	0.378	0.559	0.477	0.403	0.130	0.243	0.613	0.477	0.824

TABLE 6. Heterotrait-Monotrait Ratio of Correlation (HTMT).

	BISC	DK	EP	FC	GS	HB	IN	PC	PE	PG	PU	SC	SE	SF	SI
BISC															
DK	0.582														
EP	0.631	0.365													
FC	0.540	0.580	0.357												
GS	0.674	0.402	0.472	0.547											
HB	0.543	0.350	0.549	0.535	0.556										
IN	0.637	0.535	0.668	0.375	0.487	0.494									
PC	0.835	0.612	0.657	0.915	0.772	0.732	0.618								
PE	0.491	0.498	0.242	0.604	0.478	0.344	0.522	0.504							
PG	0.733	0.519	0.570	0.609	0.786	0.641	0.589	0.760	0.490						
PU	0.462	0.221	0.268	0.256	0.341	0.254	0.423	0.334	0.653	0.295					
SC	0.598	0.367	0.703	0.476	0.550	0.609	0.684	0.729	0.269	0.617	0.290				
SE	0.673	0.807	0.392	0.703	0.481	0.478	0.465	0.665	0.618	0.646	0.324	0.408			
SF	0.150	0.361	0.100	0.357	0.266	0.289	0.337	0.344	0.422	0.230	0.111	0.228	0.321		
SI	0.439	0.614	0.340	0.706	0.471	0.459	0.453	0.706	0.569	0.463	0.157	0.294	0.712	0.523	

constructs can be considered distinct and unrelated to each other, thus supporting the validity of the model.

#### C. STRUCTURAL MODEL ASSESSMENT

The structural assessment model employed in the current study is presented in this section, which includes the path analysis, and the assessment of model fit and effect sizes. By applying these techniques, this study aims to gain insight into the relationships between the variables. Hence, the results of the research hypotheses are visually presented in Figure 4, and further details and explanations are provided in Tables 7, 8, and 9.

#### 1) R-SQUARE AND EFFECT SIZE

In general, a low or moderate coefficient of determination (R-square;  $R^2$ ) suggests that the endogenous variables in the model have relatively weak to moderate explanatory power. Additionally, in most cases, the values of Q-square ( $Q^2$ ) obtained from blindfolding algorithms tend to be lower than the values of  $R^2$  [127]. According to the result (see Table 7), the model was able to explain a significant amount of variation in respondents' perceptions of the behavioral intention regarding IoT-based smart campus ( $R^2 = 0.632$ ). This indicates that the model accounts for 63.2% of the explanatory capability. Moreover, the research model also

showed significant results and variation for the behavioral intention of IoT-based smart campuses ( $Q^2 = 0.351$ ). These results show that the research model successfully captures and explains a considerable portion of the variance in respondents' behavioral intention related to IoT-based smart campuses. The  $Q^2$  value further indicates that the model's predictive power is meaningful, providing additional support for the research findings.

Moreover, the effect sizes of the endogenous variables for the research model are summarized in Table 7. This measures the magnitude or strength of the relationship between variables and helps determine the practical significance of the findings. According to prior studies [141], [142], effect sizes can be categorized as small, medium, or large, or very small, very large, and huge effect sizes, providing a measure of practical significance in the interpretation of research findings. Similarly, previous studies in the field of technology adoption have reported effect sizes in the range of very small to small [143], [144]. In this case, the effect sizes are categorized as very small, small, or no effect. Based on the table, most of the endogenous variables show very small effect sizes. Variables such as DK, EP, FC, GS, IN, PG, PU, SF, and SI have effect sizes ranging from 0.010 to 0.022, indicating a very small impact on the behavioral intention of IoT-based smart campuses. These



FIGURE 4. Proposed structural model of IoT-based smart campus adoption intention.

variables, while statistically significant, have relatively weak practical significance in explaining the variance in behavioral intention. On the other hand, the variable PC (privacy concern) shows a small effect size of 0.147. Although it is categorized as small, it indicates a relatively stronger impact compared to the other variables. This suggests that privacy concern has a slightly more substantial influence on behavioral intention in the context of IoT-based smart campus adoption. Additionally, the variables HB, PE, SC, and SE have effect sizes close to zero, indicating no significant impact on behavioral intention. These variables do not contribute meaningfully to the variance in behavioral intention in the context of the research model.

# 2) MODEL FIT

There are several model fit indices that were recommended in the literature to examine and assess the goodness-of-fit of the saturated model and the estimated model (Hair et al. [127] and Henseler et al. [138]) of the proposed IoT-based Smart Campus adoption model. The model fit indices include SRMR (Standardized Root Mean Square Residual), d\_ULS (d-value based on unweighted least squares), d\_G (d-value based on geodesic distances), and Chi-square, as presented in Table 8. Firstly, both the saturated model and the estimated model have the same SRMR value of 0.077. A lower SRMR

126012

value indicates a better fit, and since the SRMR values are the same for both models, it suggests that the estimated model fits the data reasonably well. Secondly, similar to the SRMR, the d\_ULS values are the same for both the saturated model and the estimated model (12.331). A lower d ULS value indicates a better fit, and since the values are the same, it suggests that the estimated model provides a comparable fit to the saturated model. Thirdly, the d G values are identical for both the saturated model and the estimated model (3.970). A lower d\_G value indicates a better fit, and since the values are the same, it suggests that the estimated model is comparable to the saturated model in terms of fit. Finally, Chi-square is a traditional measure used to evaluate the fit of a structural equation model [139]. However, it is known to be sensitive to sample size, and in large samples, even minor model misspecifications can lead to a significant Chi-square value. In this case, both the saturated model and the estimated model have the same Chi-square value of 9061.057. Hence, based on these fit indices, the estimated model provides a reasonably good fit to the data, as indicated by the comparable values to the saturated model.

# 3) HYPOTHESIS TESTING

The hypothesized relationship was assessed through the path coefficient, which is presented in Table 9. Similarly, Table 9

#### TABLE 7. PLS Measures (R Square and F Square).

		0				
Endogenous variables	R²	R <sup>2</sup> Adjusted	$Q^2$	Endogenous variables	BISC	Comment
Behavior intention of IoT-based Smart Campus adoption	0.632	0.620	0.351	DK	0.014	Very small
				EP	0.017	Very small
				FC	0.011	Very small
				GS	0.022	Very small
				HB	0.001	No effect
				IN	0.010	Very small
				PC	0.147	Small
				PE	0.002	No effect
				PG	0.020	Very small
				PU	0.022	Very small
				SC	0.000	No effect
				SE	0.038	Very small
				SF	0.022	Very small
				SI	0.013	Very small

#### TABLE 8. Model fit indices of the Proposed Model.

Coefficients	Saturated model	Estimated model
SRMR	0.077	0.077
d_ULS	12.331	12.331
d_G	3.970	3.970
Chi-square	9061.057	9061.057

also contains results for t-values and p-values, indicating that the hypothesis is examined statistically. Firstly, the result of the structural model examines the multicollinearity in the data through the variance inflation factor (VIF). Accordingly, the result observed that the VIF values are significantly less than 5. This shows that multicollinearity is not a concern in this study [127]. Remarkably, this supports the CMB results, as well as the KMO and Bartlett's test of sphericity under the normality test. Hence, it is now statistically practical to discuss the outputs obtained for the formulated research hypotheses.

The technology specifics factors that were derived from TAM have two constructs; PE and PU which are H1 and H2. The finding shows that hypothesis 1 was not supported, as there was an insignificant relationship between PE and behavioral intention of IoT-based smart campuses (t = 0.697, p = 0.486). Hypothesis 2 is supported, with a positive relationship between PU and behavioral intention of IoT-based smart campuses (t = 2.609, p = 0.009). In the organizational-specific factors, results were obtained for H3, H4, and H5, which stated that FC has a statistically significant impact on the behavioral intention of IoT-based smart campus (t = 2.157, p = 0.031), service collaboration is not significant on the behavioral intention of IoT-based smart campus (t = 0.199, p = 0.842), and propagation has a statistically significant impact on behavioral intention of IoT-based smart campus (t = 2.724, p = 0.006).

In addition, the findings of this study support all the hypotheses for the environmental specifics factors; H6, H7, and H8. The result shows that government support, social influence, and external pressure have a statistically significant impact on the behavioral intention of IoT-based smart campuses (t = 2.474, p = 0.013; t = 2.209, p = 0.027; t = 2.271, p = 0.023). In the end-user-specific

factors, six hypotheses were tested. The result shows that the impact of privacy concern, self-efficacy, and domain-specific knowledge was positive and significant on the behavioral intention of IoT-based smart campus, indicating support for H9, H11, and H12 (t = 7.714, p = 0.000; t = 3.779, p = 0.000; t = 1.981, p = 0.048), and satisfaction is negative and significant on the behavioral intention of IoT-based smart campus, indicating support for hypothesis H10 (t = 2.803, p = 0.005). However, the impact of habit and innovativeness on the behavioral intention of IoT-based smart campuses was not significant (t = 0.602, p = 0.547; t = 1.917, p = 0.0055), rejecting support for H13 and H14. Table 8 summarised the hypotheses test and the corresponding outcome.

#### D. ANN RESULT AND ANALYSIS

The ANN analysis allows for the prediction of outcomes based on complex relationships and patterns in the data. This captures non-linear relationships and interactions among variables, making it a powerful tool for forecasting and predicting smart campus adoption. By training the neural network on a dataset, the model learns the underlying patterns and makes accurate predictions on new, unseen data. Accordingly, a specific number of hidden neurons were created. These neurons were then activated using the hyperbolic tangent activation function, which is commonly used to introduce non-linearity and capture complex patterns in data. On the other hand, the output layers of the model were activated using the sigmoid activation function, which maps the output values between 0 and 1, often used in binary classification tasks. To evaluate the predictive accuracy of the ANN model, the researchers employed the root mean square error (RMSE) metric. RMSE calculates the average difference between the predicted and actual values, providing a measure of the model's performance in terms of the error between the predicted and observed data points. The calculation of RMSE was performed for each network within the ANN model. The selection of the hyperbolic tangent and sigmoid activation functions, along with the use of RMSE as an evaluation metric, is supported by previous studies (References [21], [23], and [35]). These studies likely

TABLE 9.	PLS structura	l model results	and hypotheses	test.
----------	---------------	-----------------	----------------	-------

Hypotheses	Relationships	Path Coefficients	SE	T statistics	P values	Decision
HI	PE ->BISC	0.036	0.035	0.697	0.486	Not Accepted
H2	PU ->BISC	0.114	0.116	2.609	0.009	Accepted
H3	FC ->BISC	-0.101	-0.100	2.157	0.031	Accepted
H4	SC ->BISC	0.009	0.011	0.199	0.842	Not Accepted
H5	PG ->BISC	0.136	0.141	2.724	0.006	Accepted
H6	GS ->BISC	0.123	0.120	2.474	0.013	Accepted
H7	SI ->BISC	-0.108	-0.107	2.209	0.027	Accepted
H8	EP ->BISC	0.105	0.107	2.271	0.023	Accepted
H9	PC ->BISC	0.391	0.391	7.714	0.000	Accepted
H10	SF ->BISC	-0.106	-0.103	2.803	0.005	Accepted
H11	SE ->BISC	0.200	0.198	3.779	0.000	Accepted
H12	DK ->BISC	0.112	0.107	1.981	0.048	Accepted
H13	HB ->BISC	-0.023	-0.025	0.602	0.547	Not Accepted
H14	IN ->BISC	0.087	0.088	1.917	0.055	Not Accepted

TABLE 10. Root-mean-square error (RMSE) values for training and testing.

Network	RMSE (Train-	RMSE	RMSE difference
	ing)	(Testing)	
1	0.175	0.177	0.002
2	0.151	0.176	0.025
3	0.165	0.147	0.018
4	0.142	0.186	0.043
5	0.157	0.169	0.012
6	0.145	0.165	0.020
7	0.179	0.160	0.018
8	0.162	0.175	0.013
9	0.150	0.181	0.031
10	0.140	0.149	0.010
Mean	0.157	0.169	0.019
SD	0.01351285	0.01300188	0.011685675
Innut Nour	ons PG SC IN S	E EP GS PU S	I SE PE HB DK EC an

Input Neurons: PG, SC, IN, SE, EP, GS, PU, SI, SF, PE, HB, DK, FC, and PC; SD=Standard deviation.

demonstrated the effectiveness of these choices in similar analysis scenarios or data domains. Table 10 presents the RMSE values for the ANN model, indicating an average RMSE of 0.485 for the training data and 0.499 for the testing data. These values suggest that the error is small, demonstrating the model's satisfactory prediction power. A smaller RMSE indicates a more precise fit and forecast of the data, indicating higher predictive accuracy. Furthermore, the number of hidden neurons with non-zero synaptic weights in the ANN model was utilized to estimate the significance of the factors in the model. Accordingly, Figure 5 provides a visual representation of one of the 10-fold cross-validation iterations of the ANN model based on Reference [145] guidelines.

ANN analysis allows for the identification of important features or variables that contribute significantly to the predicted outcomes. By calculating the relative importance or weights of the input variables, it provides insights into the factors that have the most influence on the dependent variable. This information can be valuable in understanding the key drivers and making informed decisions. Accordingly, after establishing the predictive relevance of the ANN model, a sensitivity analysis was conducted to assess the predictive potential of the exogenous variables in relation to the

126014

endogenous variables [35], [146]. The relative importance of each exogenous variable was calculated, and the normalized relative values were derived, as displayed in Table 11. Notably, when examining the 14 factors using the NN models, propagation emerged as the most influential predictor in the model, with a normalized relative value of 81.14%. The results also revealed that service collaboration (77.91%), innovativeness (73.23%), self-efficacy (72.12%), external pressure (71.29%), government support (60.97%), perceived usefulness (59.95%), and social influence (59.93%) were all significant predictors of IoT-based smart campus adoption, albeit with varying degrees of importance. The table presents the comprehensive relevance of each variable. Overall, ANN analysis offers a robust and versatile approach to understanding, predicting, and modeling complex systems, making it a valuable tool in various fields, including IoT-based smart campus models. It provides insights into the relationships between variables, identifies important predictors, and enables accurate predictions, ultimately enhancing decision-making and understanding of the system under investigation.

#### **V. DISCUSSION**

This study conducts empirical studies that explore the perceptions, attitudes, and behaviors of key stakeholders (such as faculty and administrators) towards IoT-based smart campus solutions. The study employs quantitative research methods (survey questionnaires) to collect data on the identified factors and their relationship with the adoption of smart campus solutions. The model was validated through statistical analysis techniques such as structural equation modeling (SEM). The study provides insights into the specific relationships and strengths of the factors influencing the behavioral intention to adopt IoT-based smart campus solutions. Specifically, 14 hypotheses were tested in this study (10 were supported and 4 were rejected); the findings of the study are discussed accordingly.

Firstly, the factors under the technology-specific factors, derived from TAM suggest that the relationship between perceived ease of use and behavior intention of IoT-based

# **IEEE**Access



FIGURE 5. ANN Model. Notes\*: Hidden layer activation function: Hyperbolic tangent; Output layer activation function: Sigmoid; Input neurons: PG, SC, IN, SE, EP, GS, PU, SI, SF, PE, HB, DK, FC, and PC.

TABLE 11. Sensitivity Analysis: Independent Variable Relative Importance.

Networks	N1	N2	N3	N4	N5	N6	N7	N8	N9	N10	Mean (AVG)	Normalized RI (AVG)
PG	0.117	0.11	0.125	0.108	0.098	0.04	0.088	0.121	0.092	0.068	0.0962	81.14%
SC	0.113	0.07	0.106	0.082	0.075	0.105	0.072	0.091	0.109	0.092	0.0913	77.91%
IN	0.153	0.09	0.064	0.082	0.059	0.115	0.078	0.102	0.068	0.065	0.0875	73.23%
SE	0.097	0.13	0.068	0.049	0.113	0.092	0.069	0.069	0.078	0.084	0.085	72.12%
EP	0.093	0.05	0.062	0.105	0.065	0.072	0.113	0.087	0.099	0.082	0.0825	71.29%
GS	0.044	0.09	0.112	0.074	0.074	0.055	0.056	0.071	0.068	0.07	0.0709	60.97%
PU	0.072	0.06	0.049	0.051	0.083	0.073	0.095	0.084	0.067	0.063	0.0699	59.95%
SI	0.065	0.08	0.053	0.083	0.07	0.065	0.058	0.082	0.045	0.096	0.0692	59.93%
SF	0.075	0.06	0.08	0.055	0.057	0.064	0.049	0.068	0.058	0.063	0.0633	53.80%
PE	0.048	0.08	0.042	0.043	0.06	0.081	0.066	0.055	0.07	0.054	0.0594	51.37%
HB	0.03	0.05	0.082	0.096	0.059	0.043	0.06	0.033	0.062	0.07	0.0586	50.99%
DK	0.044	0.06	0.075	0.062	0.066	0.064	0.068	0.036	0.041	0.068	0.058	50.35%
FC	0.024	0.05	0.034	0.061	0.062	0.068	0.079	0.055	0.075	0.055	0.0564	49.42%
PC	0.025	0.04	0.048	0.05	0.059	0.061	0.05	0.045	0.066	0.071	0.0517	45.49%

Input Neurons: PG, SC, IN, SE, EP, GS, PU, SI, SF, PE, HB, DK, FC, and PC; RI=Relative importance.

Smart Campus adoption is not statistically significant. This finding support existing studies [98], [112], [114], [147] but contradict others [81], [111], [148], [149], [150], [151], [152], [153] whose found a significant impact of ease of use. Accordingly, it can be concluded that the study did not find support for a significant relationship between perceived ease of use and behavior intention of IoT-based Smart Campus adoption. However, the study suggests that the relationship between perceived usefulness and behavior intention of IoT-based Smart Campus adoption. However, the study suggests that the relationship between perceived usefulness and behavior intention of IoT-based Smart Campus adoption is statistically significant. This finding supports existing studies on the significance of usefulness on IoT-related devices [81], [98],

[100], [111], [112], [114], [147], [149], [150], [151], [152], [153], [154], [155], [156]. Although, this relation is not supported by the work of [148]. Nevertheless, based on this study's findings, it can be concluded that the study found a significant relationship between perceived usefulness and behavior intention of IoT-based Smart Campus adoption. This suggests that individuals' perception of the usefulness of IoT-based Smart Campus technologies influences their intention to adopt and use them. These results contribute to the understanding of the factors that drive behavior intention in the context of IoT-based Smart Campus adoption.

Secondly, in organizational factors, the study suggests that the relationship between facilitating conditions and behavior intention of IoT-based Smart Campus adoption is statistically significant, which is supported by the existing literatures [86], [109], and [150]. This suggests that the availability of facilitating conditions may hinder or negatively influence individuals' intention to adopt and use IoT-based Smart Campus technologies. Overall, these findings contribute to the understanding of the role of facilitating conditions in shaping behavior intention in the context of IoT-based Smart Campus adoption. Also, the relationship between propagation and behavior intention of IoT-based Smart Campus adoption is statistically significant. Accordingly, the finding of this study emphasized the significance of propagation, as highlighted by [3], with empirical evidence. Therefore, This suggests that the extent to which information about the benefits and advantages of IoT-based Smart Campus adoption is spread and communicated (propagation) positively influences individuals' behavior and intention to adopt and use these technologies. However, the finding suggests that the relationship between service collaboration and behavioral intention of IoT-based Smart Campus adoption is not statistically significant. This finding does not support existing work regarding the influence of service collaboration for the study of IoT service orchestration in Smart Government [81]. This study finding implies that in the context of IoT-based Smart Campus adoption, service collaboration may not have a significant impact on individuals' behavior and intention to adopt and use these technologies.

Thirdly, the finding suggests that the relationship between government support, social influence, and external pressure with behavior intention of IoT-based Smart Campus adoption is statistically significant. This finding suggests that the level of support and initiatives provided by the government positively influence individuals' behavior and intention to adopt and use IoT-based Smart Campus technologies. This study collaborates with Reference [73], who found that government policy significantly influences embracing the Smart-Home revolution. Accordingly, this study implies that government policies, funding, and regulations that promote the adoption and implementation of these technologies can have a significant impact on individuals' intention to adopt and utilize them. These results highlight the role of government support in fostering the adoption of IoT-based Smart Campus technologies.

Similarly, the finding suggests that social influence, such as opinions, recommendations, and perceptions of others, has a significant impact on individuals' behavioral intention to adopt and use IoT-based Smart Campus technologies. This finding supports existing studies on the significance of social influence [109], [150] as well as subjective norm [152] on IoT adoption. However, a study by Reference [86] found social influence not significant for the IoT-Based Smart Meter adoption. According to the findings of this study, it implies that when individuals perceive a higher level of social influence against the adoption of these technologies, their behavioral intention to adopt and utilize them decreases. In addition, this finding suggests that external pressures, such as regulations, policies, or institutional mandates, play a significant role in influencing individuals' behavioral intention to adopt and use IoT-based Smart Campus technologies. This finding does not support existing literature on the influence of external pressure for a study regarding Smart factory adoption for small and medium enterprises [87]. It implies that when individuals perceive a higher level of external pressure to adopt these technologies, their behavioral intention to adopt and utilize them increases. These results highlight the importance of external factors and institutional context in shaping individuals' intention to adopt IoT-based Smart Campus technologies. Overall, the findings emphasize the importance of organizational-specific factors in shaping the adoption of IoT-based Smart Campus technologies.

Furthermore, the findings of the end-user-specific factors suggest support for privacy concerns, satisfaction, selfefficacy, and domain-specific knowledge. The finding indicates that individuals' concerns regarding privacy in the context of adopting and using IoT-based Smart Campus technologies play a significant role in shaping their behavior intention. This result supports existing literature regarding the impact of privacy and trust in a study about Google Glass adoption [149] as well concern for privacy [73], [89], [92], [96], [98], [100], [156], [157]. This highlights the importance of addressing privacy concerns when promoting the adoption of IoT-based Smart Campus technologies. Nonetheless, the findings emphasize the need to address privacy concerns and ensure data protection in the implementation of IoT-based Smart Campus technologies. Similarly, the study highlights the importance of addressing user satisfaction in the implementation of IoT-based Smart Campus technologies as is also significant in this study, which supported the existing work on the IoT technology such as near field communication (NFC) customer satisfaction and loyalty [102]. It implies that organizations and institutions should focus on enhancing user satisfaction by providing reliable, user-friendly, and efficient technologies and services. This emphasizes the need for continuous improvement, user feedback, and responsive support mechanisms to ensure high levels of user satisfaction, ultimately driving positive behavior intention and successful implementation of these technologies on campus.

In addition, the study finding indicates that individuals' self-efficacy, their belief in their own ability to adopt and use IoT-based Smart Campus technologies, has a significant impact on their behavioral intention. Similarly, this finding supports existing studies about the impact of self-efficacy for IoT-related devices [102], [109], [156]. This suggests that when individuals have higher levels of self-efficacy, they are more likely to have a positive behavior intention toward adopting and utilizing these technologies. These results highlight the importance of self-efficacy in shaping individuals' behavior intention and adoption of IoT-based Smart Campus technologies. Moreover, the finding indicates

that individuals' domain-specific knowledge, expertise, and understanding of the specific domain related to IoT-based Smart Campus technologies, have a significant impact on their behavioral intention. Similarly, this finding supports existing studies about the positive impact of this construct [110], [158]. Accordingly, the finding of this study suggests that when individuals possess higher levels of domainspecific knowledge, they are more likely to have a positive behavior intention toward adopting and utilizing smart campus technologies.

However, the relationship between habit and behavior intention of IoT-based Smart Campus adoption is not statistically significant. Likewise, the relationship between innovativeness and behavior intention of IoT-based Smart Campus adoption is not statistically significant. Specifically, the finding indicates that habit, defined as the automatic and routine behaviors individuals develop, does not play a significant role in shaping their behavior intention to adopt and utilize IoT-based Smart Campus technologies. The habit, as part of the cluster of individual factors, was found to be significant in IoT technologies adoption [92] as well as in the study of IoT-Based Smart Meter [86]. This study suggests that individuals' behavior intention is not strongly influenced by their existing habits when it comes to adopting smart campus technologies. Also, the finding of the study suggests that individuals' level of innovativeness, which refers to their willingness to adopt new technologies and embrace change, may not have a strong influence on their behavioral intention to adopt and utilize IoT-based Smart Campus technologies. This result supports existing literature regarding the insignificant influence of innovativeness on behavioral intention [95], [148]. However, some studies [73], [111] found innovativeness as a driving factor in smart home technology adoption.

Nevertheless, it is important to note that although the impact of perceived ease of use, service collaboration, habit, and innovativeness on the behavioral intention of IoT-based smart campuses are not statistically significant, it does not necessarily imply that these variables have no influence at all. Other factors and variables in the model may have a more dominant impact on behavior intention, overshadowing the role of ease of use, service collaboration, habit, and innovativeness in this specific context. Nevertheless, the findings obtained from SEM analysis support the earlier study regarding the significance of perceived ease of use, habit, and innovativeness, which have lower ranking compared to other factors [28]. Further research and analysis may be needed to explore additional variables or factors that might explain the behavior intention of IoT-based Smart Campus adoption more comprehensively. These findings suggest that perceived ease of use, service collaboration, habit, and innovativeness, may not be a strong predictor in this context, and other factors such as perceived usefulness, and other external influences may play a more prominent role in shaping individuals' behavior intention towards adopting IoT-based Smart Campus technologies.

Additionally, the analysis of predictive power validates the findings of the structural model regarding the fit of the proposed model, revealing that the overall predictive capability of the model is relatively adequate. Based on the ANN analysis, the RMSE was small, suggesting that the model demonstrates strong predictive ability. Subsequently, the sensitivity analysis conducted using ANN identified propagation as the most influential predictor in the model, followed by service collaboration and innovativeness, in this order. This order of importance associated with variables may appear contradictory to the results obtained from the path coefficient analysis, which suggested that the impact of service collaboration and innovativeness lacked significance. Nevertheless, it is worth noting that the core principle underlying ANN is to enhance the performance and learning capabilities of neural network (NN) models [33], [35]. Consequently, the utilization of ANN analysis has effectively improved the significance of other variables, potentially revealing a meaningful association between service collaboration and innovativeness on the outcome of IoT-based smart campus adoption intention.

# A. PRACTICAL IMPLICATIONS

The findings of this study suggest several important implications for university administrators and decision-makers aiming to enhance smart campus adoption. First and foremost, universities should provide adequate support to users of smart campus solutions. This support may come in the form of training programs and resources to improve users' technological abilities and familiarity with smart campus technologies, enabling them to use these solutions comfortably and effectively. Additionally, government support is crucial in promoting smart campus adoption. Governments should develop enabling policies and provide financial incentives to universities, facilitating the implementation and integration of smart campus solutions. Such support can significantly contribute to the successful adoption and implementation of smart campus initiatives.

Secondly, addressing users' privacy concerns is another vital aspect to consider. Universities and administrators should develop robust privacy policies and mechanisms to protect users' personal information and ensure compliance with relevant data protection regulations. By effectively managing privacy-related issues, universities can foster trust and confidence among users, encouraging their active participation and engagement in the smart campus environment. Moreover, promoting the use of smart campus solutions among users is essential. Universities should create awareness campaigns, highlighting the benefits and advantages of using smart campus technologies. By showcasing the positive impact and value of these solutions, users can be motivated to embrace and utilize them more readily.

Additionally, attention should be given to factors associated with users, such as self-efficacy, domain-specific knowledge, habit, and innovativeness. Efforts should be made to enhance these factors through targeted interventions,



training programs, and continuous learning opportunities. By empowering users and building their confidence, universities can drive higher levels of smart campus adoption and engagement. Lastly, universities should ensure that smart campus solutions are easily accessible and user-friendly. Quick and convenient access to these technologies enables users to fulfill their responsibilities efficiently, leading to enhanced satisfaction and overall positive experiences with the smart campus environment. By implementing these recommendations, universities can pave the way for successful smart campus adoption, fostering a technologically advanced and innovative educational environment that benefits both the institutions and their users.

# **B. POLICY IMPLICATIONS**

Based on the findings of the study, several policy recommendations can be made to promote the adoption of IoT-based smart campus solutions in higher education institutions. These recommendations aim to address the identified factors that influence adoption and create an enabling environment for the successful implementation of smart campus initiatives, as illustrated in Figure 6. Firstly, governments should provide strong support to universities in the form of policies, funding, and infrastructure development. This support can encourage universities to invest in smart campus solutions and ensure their effective implementation. Governments can establish dedicated funding programs for universities to adopt and implement smart campus technologies, as well as provide regulatory frameworks to facilitate the integration of IoT-based solutions in higher education institutions. Secondly, policy frameworks should be established to address privacy concerns related to the collection and use of personal data in smart campus environments. Clear guidelines and regulations should be developed to ensure data protection, user consent, and secure data management practices. This can help build trust among users and alleviate privacy concerns, ultimately fostering greater adoption of smart campus solutions.

Thirdly, higher education institutions should prioritize training programs and capacity-building initiatives to enhance the technological abilities of users and stakeholders. Workshops, seminars, and training sessions can be organized to educate students, faculty, and staff on the benefits and functionalities of smart campus solutions. This can improve their skills and confidence in utilizing these technologies, leading to increased adoption and usage. Moreover, collaboration among universities, industry stakeholders, and technology providers is essential to drive smart campus adoption. Policy recommendations should encourage universities to establish partnerships with technology companies, start-ups, and other relevant organizations. These collaborations can facilitate the development of innovative solutions, knowledge sharing, and resource pooling, thereby accelerating the adoption of smart campus technologies.

In addition, policy frameworks should emphasize the importance of user-centric design principles in the development and implementation of smart campus solutions. Solutions should be intuitive, user-friendly, and tailored to the specific needs and preferences of the higher education community. User involvement in the design and evaluation process should be encouraged to ensure that the solutions meet their expectations and requirements. Furthermore, it is crucial to establish mechanisms for ongoing monitoring and evaluation of smart campus initiatives. Policy recommendations should encourage universities to regularly assess the effectiveness and impact of implemented solutions. This can help identify areas for improvement, address any challenges or issues that arise, and ensure continuous enhancement of the smart campus environment. Hence, by implementing these policy recommendations, higher education institutions and policymakers can create a supportive ecosystem for the adoption of IoT-based smart campus solutions. These recommendations address key factors identified in the study and can pave the way for a more technologically advanced and innovative higher education landscape.

# **VI. CONCLUSION**

This study was conducted to examine the factors that influence the adoption of smart campus solutions. The study is important in this area to adapt and customize the factors identified in the literature to the specific context of Iraq's higher education system. The insights from the literature inform the development of a comprehensive framework for IoT-based Smart Campus. Considering the lack of existing studies on the relationship between factors of IoT-based smart campuses and the adoption of smart campus solutions in Iraq's higher education, this study proposes a theoretical foundation for Smart Campus adoption. The component of the conceptualized framework was screened through duplicate screening, thematic analysis, frequency techniques, and AHP. Furthermore, the study employed structural equation modeling based on PLS-SEM to analyze the collected data and examine the relationships between the identified factors and the adoption of smart campus solutions.

# TABLE 12. Survey instrument.

Factors	Items	Ref
	I am concerned that smart campus services are collecting too much information from me.	[160], [161]
USE: Privacy concern	I am concerned that smart campus services will use my information for other purposes.	[160], [161]
	I am concerned that smart campus services will share my information with other parties.	[160], [161]
	I am concerned that smart campus services does not protect the privacy of my information.	[160], [161]
	The smart campus collecting my personal data without my permission concerns me.	[73]
	The information gathered by smart campus can be tracked, analyzed, and misused.	[73]
	I fear using an IoT-based smart campus service due to the loss of my data and privacy.	[73]
	I am very satisfied with the manufacturer of smart campus services.	[101]
	The manufacturer of smart campus services meets my expectations.	[101]
	The manufacturer of smart campus services fits my needs/wants.	[101]
USF: Satisfaction	I am very satisfied with the smart campus services.	[102]
Cor . Satisfaction	I feel very good about the smart campus services.	[102]
	I am very happy about the smart campus services.	[102]
	I am very pleased with the smart campus services.	[102]
	I have sufficient knowledge to use the IoT-based smart campus by myself.	[73]
USF: Self-efficacy	I will be able to operate the various IoT-based devices inside a smart campus although I have never used them before.	[73]
	Adopting IoT-based smart campus services is entirely within my control.	[73]
	I have the capacity and ability to use smart campus services.	[73]
	No one can stop me from using IoT-based Smart Campus if I wish.	Developed by the study
	I have enough information about the services of IoT-based Smart Campus.	[109]
USF: Domain-specific kno	where enough information about the advantages of using IoT-based Smart Campus.	[109]
	I have enough information about the ways of opening an IoT-based Smart Campus.	[109]
	I have enough information about the operation of IoT-based Smart Campus.	Developed by the study
	I have enough information about the disadvantages of using IoT-based Smart Campus.	Developed by the study
	Choosing Smart Campus Service to access building has become auto- matic to me.	[162]
USF: Habit	Using Smart Campus to perform some tasks in the university is natural to me.	[162]

# TABLE 12. (Continued.) Survey instrument.

	When I need to access a facility, using Smart Campus services is an obvious choice for me.	[162]
	I am beginning to think that using Smart Campus service is habitual for me.	Developed by the study
	I think that I will develop a habit of using Smart Campus devices and services regularly.	Developed by the study
USF: Innovativeness	Living experiences in an IoT-based smart campus is different from a traditional campus.	[73]
	The concept of an IoT-based smart campus is unique.	[73]
	The services offered by IoT-based smart campus are new and reward- ing.	[73]
	The main purpose of Smart Campus services is clear and understand- able.	[76], [78], [163]
TSF: Ease of Use	I think that learning to use the Smart Campus system would be very easy for me.	[76], [78], [163]
	With the Smart campus system, it would be easy for me to avoid difficulty in doing some work.	[76], [78], [163]
	I would find it simple to use Smart campus solutions in my school.	[76], [78], [163]
	Instructions of Smart Campus are easy to follow.	[76], [78], [163]
	Using the Smart campus system would make me feel better about myself.	[76], [78], [164]
TSF: Usefulness	By using Smart campus solutions, I would hope to be helping my school and society.	[76], [78], [164]
	The use of Smart campus services would increase my peace of mind and I would be relaxed.	[76], [78], [164]
	I think that using Smart Campus services allows me to complete transactions more quickly.	[76], [78], [164]
	I think that using Smart campus services would enable me to conduct tasks more quickly and save time.	[76], [78], [164]
	I think that there is top management support for using IoT-based Smart Campus.	[87]
ESF: Government support	I think that there is enough employee involvement in information technology. I think that there is efficient information technology investment to support IoT-based smart campuses.	[87] [87]
	I think that the government plays a significant role in framing policies encouraging smart campus use.	[73]
	Limited campaigning and publicity about IoT smart campus result in overall ignorance among the public.	[73]
	I believe that government support contributes to successful IoT-based smart campus adoption by providing a friendly and suitable environ- ment.	[93]
	I believe that using IoT Smart campus can improve my social status.	[164]
ESF: Social influence	I think people who use IoT Smart campus have higher reputations than those who do not use IoT Smart campus.	[164]

# TABLE 12. (Continued.) Survey instrument.

	I think that using IoT smart campus will make people respect me more.	Developed by the study
	I think that people who use smart campuses are more intelligent and smart.	Developed by the study
	People will think that I am smart and intelligent when they see that I am using IoT smart campus.	Developed by the study
ESF: External pressure	I think that external pressure such as competition will drive the university to implement an IoT smart campus.	[93]
	I think that IoT smart campuses will help universities maintain com- petitiveness with other universities.	[93]
	I think that prospective students will make universities implement IoT- based smart campuses.	Developed by the study
	I have the necessary resources to use IoT-based smart campus services.	[78]
	There is an availability of a specific person (group) to assist with IoT- based smart campus usage difficulties.	[78]
OSF: Facilitating condition	I have the necessary knowledge and information to use IoT-based smart campus.	[78]
	IoT-based smart campus is compatible with other technologies I use.	[78]
	The internet facilities I have fully meet the needs to use IoT-based smart campus.	Developed by the study
	I have adequate and effective workshops and training to familiarize myself with the IoT-based smart campus.	Developed by the study
	My university promotes the use of IoT-based smart campuses sufficiently.	Developed by the study
	Support services and infrastructure give me more confidence to use IoT-based smart campus.	Developed by the study
	I think that human involvement is purposeful toward individual value creation regarding IoT-based smart campus services goal attainment.	[81]
OSF: Service collaboration	I think that service collaboration between government and private agencies is important for continued business success for IoT-based	[81]
	I think that service collaboration is strongly committed to a shared mission for IoT-based smart campus.	[81]
	I think that citizens are involved in government operations by actively evaluating the available information and notifying of shortcomings of IoT-based smart campuses.	[81]
	I think that replication will enable many universities to implement smart campuses.	Developed by the study
OSF: Propagation	I think that IoT-smart campus solutions can be replicated conveniently.	Developed
	I think that the cost of replicating smart campus solutions is reduced significantly.	by the study Developed by the study
	I think that the ability to propagate IoT-based smart campus services will make smart campus widely implemented.	Developed by the study
	I think that replicating smart campus solutions is easier than installing new solutions.	Developed by the study
	I will continue to use IoT smart campus rather than any conventional system.	[81]

#### TABLE 12. (Continued.) Survey instrument.

I intend to continue using IoT smart campus if a positive commitment is shown by the government towards the IoT smart campus.	[81]
If the services offered by the government are trustworthy, I will be using the IoT smart campus more precisely.	[81]
My participation in IoT smart campus is a way to respond to govern- ment calls to action regarding service innovation.	[81]
I am likely to access a university facility with an IoT smart campus system.	[165]
In my daily life, I will continue to use IoT smart campus services.	[164]

The findings of this study provide insights into the specific factors that influence the adoption of smart campus solutions in Iraq's higher education context. Specifically, significant factors include perceived usefulness, government support, social influence, privacy concern, facilitating conditions, satisfaction, self-efficacy, external pressure, propagation, and domain-specific knowledge. However, factors such as perceived ease of use, service collaboration, habit, and innovativeness were not significant based on the findings of this study. Nevertheless, this study is not without limitations. The model fit reported in this study is based on PLS-SEM model fit indices. Since CB-SEM was not utilized in this study, the model fit indices for CB-SEM (TLI, RMR, GFI, AGFI, CFI, or RMSEA) could not be reported. Consequently, it raises the question of why the researcher did not use CB-SEM in the first place to estimate and evaluate the model fit when assuming common factor models for all constructs in the PLS path model. It is important to note that the literature on PLS-SEM about model fit differs from CB-SEM, which is a full information method, while PLS-SEM is not [159]. Therefore, reporting CB-SEM model fits (TLI, RMR, GFI, AGFI, CFI, or RMSEA) is not considered in this study. As a result, future research could replicate the study using a different method, particularly CB-SEM, to analyze the model.

# **APPENDIX A**

Refer to Table 12.

# REFERENCES

- [1] R. Setiawan, M. M. V. Devadass, R. Rajan, D. K. Sharma, N. P. Singh, K. Amarendra, R. K. R. Ganga, R. R. Manoharan, V. Subramaniyaswamy, and S. Sengan, "IoT based virtual E-learning system for sustainable development of smart cities," *J. Grid Comput.*, vol. 20, no. 3, p. 24, Sep. 2022.
- [2] F. Oliveira, D. Nery, D. G. Costa, I. Silva, and L. Lima, "A survey of technologies and recent developments for sustainable smart cycling," *Sustainability*, vol. 13, no. 6, p. 3422, Mar. 2021.
- [3] N. Min-Allah and S. Alrashed, "Smart campus—A sketch," Sustain. Cities Soc., vol. 59, Aug. 2020, Art. no. 102231.
- [4] B. Ahmad, M. Umar, M. Tanko, S. Tenuche, A. Sambo, A. Abdulsalami, and A. Abdulrahim, "An IoT based smart campus architecture for institutions in developing countries," *i-Manager's J. Embedded Syst.*, vol. 7, no. 1, p. 18, 2018.
  - 701. 7, 110. 1, p. 1

- [5] W. Muhamad, N. B. Kurniawan, Suhardi, and S. Yazid, "Smart campus features, technologies, and applications: A systematic literature review," in *Proc. Int. Conf. Inf. Technol. Syst. Innov. (ICITSI)*, Oct. 2017, pp. 384–391.
- [6] Y. Atif, S. S. Mathew, and A. Lakas, "Building a smart campus to support ubiquitous learning," *J. Ambient Intell. Humanized Comput.*, vol. 6, no. 2, pp. 223–238, Apr. 2015.
- [7] A. Adamkó, "Building smart university using innovative technology and architecture," in *Smart Universities: Concepts, Systems, and Technologies*, vol. 4. Cham, Switzerland: Springer, 2017, pp. 161–188.
- [8] O. A. Amodu, U. A. Bukar, R. A. R. Mahmood, C. Jarray, and M. Othman, "Age of information minimization in UAV-aided data collection for WSN and IoT applications: A systematic review," *J. Netw. Comput. Appl.*, vol. 216, Jul. 2023, Art. no. 103652.
- [9] O. Amodu, R. Nordin, C. Jarray, U. Bukar, R. R. Mahmood, and M. Othman, "A survey on the design aspects and opportunities in age-aware UAV-aided data collection for sensor networks and Internet of Things applications," *Drones*, vol. 7, no. 4, p. 260, Apr. 2023.
- [10] L. Gupta, R. Jain, and G. Vaszkun, "Survey of important issues in UAV communication networks," *IEEE Commun. Surveys Tuts.*, vol. 18, no. 2, pp. 1123–1152, 2nd Quart., 2016.
- [11] S. A. H. Mohsan, M. A. Khan, F. Noor, I. Ullah, and M. H. Alsharif, "Towards the unmanned aerial vehicles (UAVs): A comprehensive review," *Drones*, vol. 6, no. 6, p. 147, Jun. 2022.
- [12] H. Shakhatreh, A. H. Sawalmeh, A. Al-Fuqaha, Z. Dou, E. Almaita, I. Khalil, N. S. Othman, A. Khreishah, and M. Guizani, "Unmanned aerial vehicles (UAVs): A survey on civil applications and key research challenges," *IEEE Access*, vol. 7, pp. 48572–48634, 2019.
- [13] S. M. Lee, D. Lee, and Y. S. Kim, "The quality management ecosystem for predictive maintenance in the industry 4.0 era," *Int. J. Quality Innov.*, vol. 5, no. 1, pp. 1–11, Dec. 2019.
- [14] A. Eltawil, N. Mostafa, and Y. Matsushita, "Toward a smart and sustainable campus: Future vision, opportunities, and challenges," in *Resilient* and Responsible Smart Cities, vol. 1. Cham, Switzerland: Springer, 2021, pp. 233–242.
- [15] A. Petcovici and E. Stroulia, "Location-based services on a smart campus: A system and a study," in *Proc. IEEE 3rd World Forum Internet Things (WF-IoT)*, Dec. 2016, pp. 94–99.
- [16] M. Turner, B. Kitchenham, P. Brereton, S. Charters, and D. Budgen, "Does the technology acceptance model predict actual use? A systematic literature review," *Inf. Softw. Technol.*, vol. 52, no. 5, pp. 463–479, May 2010.
- [17] R. Sneesl, Y. Y. Jusoh, M. A. Jabar, and S. Abdullah, "Revising technology adoption factors for IoT-based smart campuses: A systematic review," *Sustainability*, vol. 14, no. 8, p. 4840, Apr. 2022.
- [18] R. Sneesl, Y. Y. Jusoh, M. A. Jabar, and S. Abdullah, "Conceptualizing IoT-based smart campus adoption model for higher education institutions: A systematic literature review," in *Proc. Appl. Informat. Int. Conf. (AiIC)*, May 2022, pp. 7–12.
- [19] V. Venkatesh, J. R., Y. Y., M. Jabar, and S. Abdullah, "Adoption and use of AI tools: A research agenda grounded in UTAUT," *Ann. Oper. Res.*, vol. 308, p. 4840, 2022.

- [20] G. W.-H. Tan, K.-B. Ooi, L.-Y. Leong, and B. Lin, "Predicting the drivers of behavioral intention to use mobile learning: A hybrid SEMneural networks approach," *Comput. Hum. Behav.*, vol. 36, pp. 198–213, Jul. 2014.
- [21] J.-J. Hew, M. N. B. A. Badaruddin, and M. K. Moorthy, "Crafting a smartphone repurchase decision making process: Do brand attachment and gender matter?" *Telematics Informat.*, vol. 34, no. 4, pp. 34–56, Jul. 2017.
- [22] J.-J. Hew, L.-Y. Leong, G. W.-H. Tan, V.-H. Lee, and K.-B. Ooi, "Mobile social tourism shopping: A dual-stage analysis of a multi-mediation model," *Tourism Manage.*, vol. 66, pp. 121–139, Jun. 2018.
- [23] Y. Li, S. Yang, S. Zhang, and W. Zhang, "Mobile social media use intention in emergencies among gen y in China: An integrative framework of gratifications, task-technology fit, and media dependency," *Telematics Informat.*, vol. 42, Sep. 2019, Art. no. 101244.
- [24] U. A. Bukar, F. Sidi, M. A. Jabar, R. N. H. B. Nor, S. Abdullah, and I. Ishak, "A multistage analysis of predicting public resilience of impactful social media crisis communication in flooding emergencies," *IEEE Access*, vol. 10, pp. 57266–57282, 2022.
- [25] V. Ahmed, K. A. Alnaaj, and S. Saboor, "An investigation into stakeholders' perception of smart campus criteria: The American university of Sharjah as a case study," *Sustainability*, vol. 12, no. 12, p. 5187, Jun. 2020.
- [26] A. A. Osuwa, J. O. Katende, and A. A. Osuwa, "Perception of smart campus big data analytics in university," in *Proc. 3rd World Conf. Smart Trends Syst. Secur. Sustainability (WorldS)*, Jul. 2019, pp. 48–58.
- [27] O. Pribyl, S. Opasanon, and T. Horák, "Student perception of smart campus: A case study of Czech republic and Thailand," in *Proc. Smart City Symp. Prague (SCSP)*, May 2018, pp. 1–7.
- [28] R. Sneesl, Y. Y. Jusoh, M. A. Jabar, S. Abdullah, and U. A. Bukar, "Factors affecting the adoption of IoT-based smart campus: An investigation using analytical hierarchical process (AHP)," *Sustainability*, vol. 14, no. 14, p. 8359, Jul. 2022.
- [29] Z. Y. Dong, Y. Zhang, C. Yip, S. Swift, and K. Beswick, "Smart campus: Definition, framework, technologies, and services," *IET Smart Cities*, vol. 2, no. 1, pp. 43–54, Mar. 2020.
- [30] R. V. Imbar, S. H. Supangkat, and A. Z. R. Langi, "Smart campus model: A literature review," in *Proc. Int. Conf. ICT Smart Soc. (ICISS)*, Nov. 2020, pp. 1–7.
- [31] M. W. Sari, P. W. Ciptadi, and R. H. Hardyanto, "Study of smart campus development using Internet of Things technology," *IOP Conf. Ser., Mater. Sci. Eng.*, vol. 190, no. 1, 2017, Art. no. 012032.
- [32] X. Xu, Y. Wang, and S. Yu, "Teaching performance evaluation in smart campus," *IEEE Access*, vol. 6, pp. 77754–77766, 2018.
- [33] P.-C. Chang, J.-J. Lin, and W.-Y. Dzan, "Forecasting of manufacturing cost in mobile phone products by case-based reasoning and artificial neural network models," *J. Intell. Manuf.*, vol. 23, no. 3, pp. 517–531, Jun. 2012.
- [34] A. Y.-L. Chong, "Predicting m-commerce adoption determinants: A neural network approach," *Expert Syst. Appl.*, vol. 40, no. 2, pp. 523–530, Feb. 2013.
- [35] A. Teo, G. Tan, K. Ooi, T. Hew, and K. Yew, "The effects of convenience and speed in m-payment," *Ind. Manag. Data Syst.*, vol. 115, no. 2, pp. 311–331, 2015.
- [36] O. Sohaib, W. Hussain, M. Asif, M. Ahmad, and M. Mazzara, "A PLS-SEM neural network approach for understanding cryptocurrency adoption," *IEEE Access*, vol. 8, pp. 13138–13150, 2020.
- [37] G. Wang, G. W.-H. Tan, Y. Yuan, K.-B. Ooi, and Y. K. Dwivedi, "Revisiting TAM2 in behavioral targeting advertising: A deep learningbased dual-stage SEM-ANN analysis," *Technol. Forecasting Social Change*, vol. 175, Feb. 2022, Art. no. 121345.
- [38] M. Tsourela and D.-M. Nerantzaki, "An Internet of Things (IoT) acceptance model. Assessing Consumer's behavior toward IoT products and applications," *Future Internet*, vol. 12, no. 11, p. 191, Nov. 2020, doi: 10.3390/fi12110191.
- [39] A. M. Al-Momani, M. A. Mahmoud, and M. S. Ahmad, "Factors that influence the acceptance of Internet of Things services by customers of telecommunication companies in Jordan," *J. Org. End User Comput.*, vol. 30, no. 4, pp. 51–63, Oct. 2018, doi: 10.4018/joeuc.2018100104.
- [40] D. Pal, X. Zhang, and S. Siyal, "Prohibitive factors to the acceptance of Internet of Things (IoT) technology in society: A smart-home context using a resistive modelling approach," *Technol. Soc.*, vol. 66, Aug. 2021, Art. no. 101683, doi: 10.1016/j.techsoc.2021.101683.

- [41] A. Rey, E. Panetti, R. Maglio, and M. Ferretti, "Determinants in adopting the Internet of Things in the transport and logistics industry," *J. Bus. Res.*, vol. 131, pp. 584–590, Jul. 2021, doi: 10.1016/j.jbusres.2020.12.049.
- [42] M. Q. Aldossari and A. Sidorova, "Consumer acceptance of Internet of Things (IoT): Smart home context," *J. Comput. Inf. Syst.*, vol. 60, no. 6, pp. 507–517, Nov. 2020, doi: 10.1080/08874417.2018.1543000.
- [43] F. Wu, Q. Zheng, F. Tian, Z. Suo, Y. Zhou, K.-M. Chao, M. Xu, N. Shah, J. Liu, and F. Li, "Supporting poverty-stricken college students in smart campus," *Future Gener. Comput. Syst.*, vol. 111, pp. 599–616, Oct. 2020, doi: 10.1016/j.future.2019.09.017.
- [44] Y.-C. Chang and Y.-H. Lai, "Campus edge computing network based on IoT street lighting nodes," *IEEE Syst. J.*, vol. 14, no. 1, pp. 164–171, Mar. 2020.
- [45] L. Luo, "Data acquisition and analysis of smart campus based on wireless sensor," *Wireless Pers. Commun.*, vol. 102, no. 4, pp. 2897–2911, Oct. 2018.
- [46] S. Fortes, J. Santoyo-Ramón, D. Palacios, E. Baena, R. Mora-García, M. Medina, P. Mora, and R. Barco, "The campus as a smart city: University of Málaga environmental, learning, and research approaches," *Sensors*, vol. 19, no. 6, p. 1349, Mar. 2019.
- [47] W. Villegas-Ch, X. Palacios-Pacheco, and S. Luján-Mora, "Application of a smart city model to a traditional university campus with a big data architecture: A sustainable smart campus," *Sustainability*, vol. 11, no. 10, p. 2857, May 2019.
- [48] Y.-B. Lin and S.-L. Chou, "SpecTalk: Conforming IoT implementations to sensor specifications," *Sensors*, vol. 21, no. 16, p. 5260, Aug. 2021.
- [49] L. C. Tagliabue, F. Re Cecconi, S. Rinaldi, and A. L. C. Ciribini, "Data driven indoor air quality prediction in educational facilities based on IoT network," *Energy Buildings*, vol. 236, Apr. 2021, Art. no. 110782.
- [50] N. Khamis and K. K. K. Li, "User experience evaluation for a bus tracking apps in smart campus initiative," *Bull. Electr. Eng. Informat.*, vol. 10, no. 4, pp. 2254–2262, Aug. 2021.
- [51] A. Yang, Q. Zhang, Y. Liu, and J. Zhao, "The improvement of DV-hop model and its application in the security performance of smart campus," *Mathematics*, vol. 10, no. 15, p. 2663, Jul. 2022.
- [52] T. Anagnostopoulos, P. Kostakos, I. Salmon, Y. Psaromiligkos, K. Ntalianis, and S. R. J. Ramson, "Spatiotemporal authentication system architecture for smart campus safety," in *Proc. 4th Int. Conf. Smart Sensors Appl. (ICSSA)*, Jul. 2022, pp. 155–160.
- [53] V. Ahmed, M. F. Khatri, Z. Bahroun, and N. Basheer, "Optimizing smart campus solutions: An evidential reasoning decision support tool," *Smart Cities*, vol. 6, no. 5, pp. 2308–2346, Sep. 2023.
- [54] A. H. M. Shabli, M. M. Rejab, N. H. Z. Abai, and S. C. Chit, "Campus bus tracking system using Lora technology: IoT system testing," in *Proc. 12th Int. Conf. Softw. Comput. Appl.*, Feb. 2023, pp. 234–239.
- [55] S. Alrashed, "Key performance indicators for smart campus and microgrid," Sustain. Cities Soc., vol. 60, Sep. 2020, Art. no. 102264.
- [56] K. AbuAlnaaj, V. Ahmed, and S. Saboor, "A strategic framework for smart campus," in *Proc. Int. Conf. Ind. Eng. Oper. Manag.*, vol. 22, 2020, pp. 790–798.
- [57] L. Yuan and J. Du, "Research on intelligent financial information framework and smarter university campus," *Int. J. Knowl.-Based Develop.*, vol. 12, nos. 3–4, p. 409, 2022.
- [58] I. Agbehadji, R. Millham, B. Awuzie, and A. Ngowi, "Stakeholder's perspective of digital technologies and platforms towards smart campus transition: Challenges and prospects," in *Proc. Int. Conf. Inform. Intell. Appl.* Cham, Switzerland: Springer, pp. 197–213.
- [59] P. I. Silva-da-Nóbrega, A. F. Chim-Miki, and M. Castillo-Palacio, "A smart campus framework: Challenges and opportunities for education based on the sustainable development goals," *Sustainability*, vol. 14, no. 15, p. 9640, Aug. 2022.
- [60] B. Valks, M. H. Arkesteijn, A. Koutamanis, and A. C. den Heijer, "Towards a smart campus: Supporting campus decisions with Internet of Things applications," *Building Res. Inf.*, vol. 49, no. 1, pp. 1–20, Jan. 2021.
- [61] D. Rico-Bautista, Y. Medina-Cárdenas, G. Maestre-Góngora, E. Barrientos-Avendaño, L. Coronel-Rojas, F. Cuesta-Quintero, and R. García-León, "Smart university: Strategic map since the adoption of technology," *RISTI-Revista Iberica de Sistemas e Tecnologias de Informacao*, vol. 2020, no. E28, pp. 711–724, 2020.
- [62] K. Polin, T. Yigitcanlar, M. Limb, and T. Washington, "The making of smart campus: A review and conceptual framework," *Buildings*, vol. 13, no. 4, p. 891, Mar. 2023.

- [63] S. D. Nagowah, H. B. Sta, and B. A. Gobin-Rahimbux, "Towards achieving semantic interoperability in an IoT-enabled smart campus," in *Proc. IEEE Int. Smart Cities Conf. (ISC)*, Oct. 2019, pp. 593–598.
- [64] V. Ahmed, S. Saboor, N. Ahmad, and M. Ghaly, "A multi-attribute utility decision support tool for a smart campus—UAE as a case study," *Frontiers Built Environ.*, vol. 8, Dec. 2022, Art. no. 1044646.
- [65] N. Samancioglu and S. Nuere, "A determination of the smartness level of university campuses: The smart availability scale (SAS)," *J. Eng. Appl. Sci.*, vol. 70, no. 1, pp. 1–19, Dec. 2023.
- [66] A. Zhamanov, Z. Sakhiyeva, R. Suliyev, and Z. Kaldykulova, "IoT smart campus review and implementation of IoT applications into education process of university," in *Proc. 13th Int. Conf. Electron., Comput. Comput. (ICECCO)*, Nov. 2017, pp. 1–4.
- [67] B. Awuzie, A. B. Ngowi, T. Omotayo, L. Obi, and J. Akotia, "Facilitating successful smart campus transitions: A systems thinking-SWOT analysis approach," *Appl. Sci.*, vol. 11, no. 5, p. 2044, Feb. 2021.
- [68] S. K. Mahariya, A. Kumar, R. Singh, A. Gehlot, S. V. Akram, B. Twala, M. I. Iqbal, and N. Priyadarshi, "Smart campus 4.0: Digitalization of university campus with assimilation of industry 4.0 for innovation and sustainability," *J. Adv. Res. Appl. Sci. Eng. Technol.*, vol. 32, no. 1, pp. 120–138, Aug. 2023.
- [69] H. Mehmood, A. Khalid, P. Kostakos, E. Gilman, and S. Pirttikangas, "A novel edge architecture and solution for detecting concept drift in smart environments," *Future Gener. Comput. Syst.*, vol. 150, pp. 127–143, Jan. 2024.
- [70] M. Sharma, R. Gupta, and P. Acharya, "Prioritizing the critical factors of cloud computing adoption using multi-criteria decision-making techniques," *Global Bus. Rev.*, vol. 21, no. 1, pp. 142–161, Feb. 2020.
- [71] F. Lanzini, J. Ubacht, and J. Greeff, "Blockchain adoption factors for SMEs in supply chain management," J. Supply Chain Manage. Sci., vol. 2, nos. 1–2, pp. 47–68, 2021.
- [72] R. Paramita and M. Dachyar, "The alternative selection for Internet of Things (IoT) implementation in medical rehabilitation," *Int. J. Adv. Sci. Technol.*, vol. 29, no. 7s, pp. 3632–3640, 2020.
- [73] D. Pal, B. Papasratorn, W. Chutimaskul, and S. Funilkul, "Embracing the smart-home revolution in Asia by the elderly: An end-user negative perception modeling," *IEEE Access*, vol. 7, pp. 38535–38549, 2019.
- [74] J. Baker, "The technology-organization-environment framework," Information Systems Theory: Explaining and Predicting Our Digital Society, vol. 1. Cham, Switzerland: Springer, 2012, pp. 231–245.
- [75] L. G. Tornatzky, M. Fleischer, and A. K. Chakrabarti, Processes of Technological Innovation. Lanham, MD, USA: Lexington Books, 1990.
- [76] F. D. Davis, "Perceived usefulness, perceived ease of use, and user acceptance of information technology," *MIS Quart.*, vol. 13, no. 3, pp. 319–340, Sep. 1989.
- [77] B. Sivathanu, "Adoption of Internet of Things (IoT) based wearables for healthcare of older adults—A behavioural reasoning theory (BRT) approach," *J. Enabling Technol.*, vol. 12, no. 4, pp. 169–185, Nov. 2018.
- [78] V. Venkatesh, J. Thong, and X. Xu, "Consumer acceptance and use of information technology: Extending the unified theory of acceptance and use of technology," *MIS Quart.*, vol. 36, no. 1, pp. 157–178, 2012.
- [79] I. Ajzen, "The theory of planned behavior," Org. Behav. Hum. Decis. Processes, vol. 50, no. 2, pp. 179–211, 1991.
- [80] S. Hong, J. Y. L. Thong, and K. Y. Tam, "Understanding continued information technology usage behavior: A comparison of three models in the context of mobile Internet," *Decis. Support Syst.*, vol. 42, no. 3, pp. 1819–1834, Dec. 2006.
- [81] S. R. Chohan and G. Hu, "Success factors influencing citizens' adoption of IoT service orchestration for public value creation in smart government," *IEEE Access*, vol. 8, pp. 208427–208448, 2020.
- [82] N. Niknejad, A. R. C. Hussin, I. Ghani, and F. A. Ganjouei, "A confirmatory factor analysis of the behavioral intention to use smart wellness wearables in Malaysia," *Universal Access Inf. Soc.*, vol. 19, no. 3, pp. 633–653, Aug. 2020.
- [83] D. Pal, C. Arpnikanondt, S. Funilkul, and W. Chutimaskul, "The adoption analysis of voice-based smart IoT products," *IEEE Internet Things J.*, vol. 7, no. 11, pp. 10852–10867, Nov. 2020.
- [84] V. Venkatesh, M. Morris, G. Davis, and F. Davis, "User acceptance of information technology: Toward a unified view," *MIS Quart.*, vol. 27, no. 3, pp. 425–478, 2003.
- [85] K. Sohn and O. Kwon, "Technology acceptance theories and factors influencing artificial intelligence-based intelligent products," *Telematics Informat.*, vol. 47, Apr. 2020, Art. no. 101324.

- [86] G. A. Alkawsi, N. Ali, and Y. Baashar, "An empirical study of the acceptance of IoT-based smart meter in malaysia: The effect of electricitysaving knowledge and environmental awareness," *IEEE Access*, vol. 8, pp. 42794–42804, 2020.
- [87] J. Y. Won and M. J. Park, "Smart factory adoption in small and medium-sized enterprises: Empirical evidence of manufacturing industry in Korea," *Technol. Forecasting Social Change*, vol. 157, Aug. 2020, Art. no. 120117.
- [88] M. Tu, "An exploratory study of Internet of Things (IoT) adoption intention in logistics and supply chain management," *Int. J. Logistics Manage.*, vol. 29, no. 1, pp. 131–151, Feb. 2018.
- [89] C.-L. Hsu and J. C.-C. Lin, "An empirical examination of consumer adoption of Internet of Things services: Network externalities and concern for information privacy perspectives," *Comput. Hum. Behav.*, vol. 62, pp. 516–527, Sep. 2016.
- [90] A. Manfreda, K. Ljubi, and A. Groznik, "Autonomous vehicles in the smart city era: An empirical study of adoption factors important for millennials," *Int. J. Inf. Manage.*, vol. 58, Jun. 2021, Art. no. 102050.
- [91] Y.-K. Lee, J.-H. Park, N. Chung, and A. Blakeney, "A unified perspective on the factors influencing usage intention toward mobile financial services," J. Bus. Res., vol. 65, no. 11, pp. 1590–1599, Nov. 2012.
- [92] O. Enaizan, B. Eneizan, M. Almaaitah, A. T. Al-Radaideh, and A. M. Saleh, "Effects of privacy and security on the acceptance and usage of EMR: The mediating role of trust on the basis of multiple perspectives," *Informat. Med. Unlocked*, vol. 21, 2020, Art. no. 100450.
- [93] D. Lin, C. K. M. Lee, and W. C. Tai, "Application of interpretive structural modelling for analyzing the factors of IoT adoption on supply chains in the Chinese agricultural industry," in *Proc. IEEE Int. Conf. Ind. Eng. Eng. Manage. (IEEM)*, Dec. 2017, pp. 1347–1351.
- [94] A. AlHogail, "Improving IoT technology adoption through improving consumer trust," *Technologies*, vol. 6, no. 3, p. 64, Jul. 2018.
- [95] M. I. Hossain, A. F. Yusof, A. R. C. Hussin, N. A. Lahad, and A. S. Sadiq, "Factors influencing adoption model of continuous glucose monitoring devices for Internet of Things healthcare," *Internet Things*, vol. 15, Sep. 2021, Art. no. 100353.
- [96] Y. Jung and J. Park, "An investigation of relationships among privacy concerns, affective responses, and coping behaviors in location-based services," *Int. J. Inf. Manage.*, vol. 43, pp. 15–24, Dec. 2018.
- [97] M. N. Alraja, M. M. J. Farooque, and B. Khashab, "The effect of security, privacy, familiarity, and trust on users' attitudes toward the use of the IoTbased healthcare: The mediation role of risk perception," *IEEE Access*, vol. 7, pp. 111341–111354, 2019.
- [98] D. Dhagarra, M. Goswami, and G. Kumar, "Impact of trust and privacy concerns on technology acceptance in healthcare: An Indian perspective," *Int. J. Med. Informat.*, vol. 141, Sep. 2020, Art. no. 104164.
- [99] C. Park, Y. Kim, and M. Jeong, "Influencing factors on risk perception of IoT-based home energy management services," *Telematics Informat.*, vol. 35, no. 8, pp. 2355–2365, Dec. 2018.
- [100] R. El-Haddadeh, V. Weerakkody, M. Osmani, D. Thakker, and K. K. Kapoor, "Examining citizens' perceived value of Internet of Things technologies in facilitating public sector services engagement," *Government Inf. Quart.*, vol. 36, no. 2, pp. 310–320, Apr. 2019.
- [101] M.-K. Kim, S. F. Wong, Y. Chang, and J.-H. Park, "Determinants of customer loyalty in the Korean smartphone market: Moderating effects of usage characteristics," *Telematics Informat.*, vol. 33, no. 4, pp. 936–949, Nov. 2016.
- [102] H. Han, A. Park, N. Chung, and K. J. Lee, "A near field communication adoption and its impact on expo visitors' behavior," *Int. J. Inf. Manage.*, vol. 36, no. 6, pp. 1328–1339, Dec. 2016.
- [103] S. Petter, W. DeLone, and E. R. McLean, "Information systems success: The quest for the independent variables," *J. Manage. Inf. Syst.*, vol. 29, no. 4, pp. 7–62, Apr. 2013.
- [104] Ö. E. Kurt, "Examining an e-learning system through the lens of the information systems success model: Empirical evidence from Italy," *Educ. Inf. Technol.*, vol. 24, no. 2, pp. 1173–1184, Mar. 2019.
- [105] H. B. Seta, T. Wati, A. Muliawati, and A. N. Hidayanto, "E-learning success model: An extention of DeLone & McLean IS' success model," *Indonesian J. Electr. Eng. Informat.*, vol. 6, no. 3, pp. 281–291, Sep. 2018.
- [106] M. Awais, M. Malik, and A. Qaisar, "A review: The job satisfaction act as mediator between spiritual intelligence and organizational commitment," *Int. Rev. Manage. Marketing*, vol. 5, no. 4, pp. 203–210, 2015.
- [107] S. Rahi, M. Ghani, and A. Ngah, "Factors propelling the adoption of Internet banking: The role of e-customer service, website design, brand image and customer satisfaction," *Int. J. Bus. Inf. Syst.*, vol. 33, no. 4, pp. 549–569, 2020.

- [108] R. Kumar, A. Sachan, and R. Kumar, "The impact of service delivery system process and moderating effect of perceived value in Internet banking adoption," *Australas. J. Inf. Syst.*, vol. 24, pp. 1–22, Jan. 2020.
- [109] M. A. Y. Yamin and B. A. Alyoubi, "Adoption of telemedicine applications among Saudi citizens during COVID-19 pandemic: An alternative health delivery system," *J. Infection Public Health*, vol. 13, no. 12, pp. 1845–1855, Dec. 2020.
- [110] Y.-S. Kao, K. Nawata, and C.-Y. Huang, "An exploration and confirmation of the factors influencing adoption of IoT-based wearable fitness trackers," *Int. J. Environ. Res. Public Health*, vol. 16, no. 18, p. 3227, Sep. 2019.
- [111] S. Nikou, "Factors driving the adoption of smart home technology: An empirical assessment," *Telematics Informat.*, vol. 45, Dec. 2019, Art. no. 101283.
- [112] J. Choi and S. Kim, "Is the smartwatch an IT product or a fashion product? A study on factors affecting the intention to use smartwatches," *Comput. Hum. Behav.*, vol. 63, pp. 777–786, Oct. 2016.
- [113] T. Saaty, What is the Analytic Hierarchy Process? Berlin, Germany: Springer, 1988.
- [114] F. Caffaro, M. M. Cremasco, M. Roccato, and E. Cavallo, "Drivers of farmers' intention to adopt technological innovations in Italy: The role of information sources, perceived usefulness, and perceived ease of use," *J. Rural Stud.*, vol. 76, pp. 264–271, May 2020.
- [115] B. Kitchenham et al., "Guidelines for performing systematic literature reviews in software engineering," Tech. Rep., Jul. 2007.
- [116] E. G. Hansen and S. Schaltegger, "The sustainability balanced scorecard: A systematic review of architectures," J. Bus. Ethics, vol. 133, no. 2, pp. 193–221, Jan. 2016.
- [117] U. A. Bukar, M. A. Jabar, F. Sidi, R. N. H. B. Nor, S. Abdullah, and M. Othman, "Crisis informatics in the context of social media crisis communication: Theoretical models, taxonomy, and open issues," *IEEE Access*, vol. 8, pp. 185842–185869, 2020.
- [118] E. Carrillo-Álvarez, E. Villalonga-Olives, J. Riera-Romaní, and I. Kawachi, "Development and validation of a questionnaire to measure family social capital," *SSM-Population Health*, vol. 8, Aug. 2019, Art. no. 100453.
- [119] S. Roopa and M. Rani, "Questionnaire designing for a survey," J. Indian Orthodontic Soc., vol. 46, no. 4, pp. 273–277, Oct. 2012.
- [120] E. Van Teijlingen and V. Hundley, "The importance of pilot studies," Social Res. Update, no. 35, pp. 1–4, 2001.
- [121] S. Vogel and J. Draper-Rodi, "The importance of pilot studies, how to write them and what they mean," *Int. J. Osteopathic Med.*, vol. 23, pp. 2–3, Mar. 2017.
- [122] P. E. Shrout and N. Bolger, "Mediation in experimental and nonexperimental studies: New procedures and recommendations," *Psychol. Methods*, vol. 7, no. 4, pp. 422–445, 2002.
- [123] D. G. Bonett and T. A. Wright, "Cronbach's alpha reliability: Interval estimation, hypothesis testing, and sample size planning," J. Org. Behav., vol. 36, no. 1, pp. 3–15, Jan. 2015.
- [124] H. Karnincic, M. Cavala, and N. Rogulj, "The relationship between handball players and alcohol and smoking habits," *J. Hum. Kinetics*, vol. 63, no. 1, pp. 127–136, Sep. 2018.
- [125] L. Kane and A. R. Ashbaugh, "Simple and parallel mediation: A tutorial exploring anxiety sensitivity, sensation seeking, and gender," *Quant. Methods for Psychol.*, vol. 13, no. 3, pp. 148–165, Oct. 2017.
- [126] A. F. Hayes and J. J. Coutts, "Use Omega rather than Cronbach's alpha for estimating reliability. But...," *Commun. Methods Measures*, vol. 14, no. 1, pp. 1–24, Jan. 2020.
- [127] J. F. Hair, J. J. Risher, M. Sarstedt, and C. M. Ringle, "When to use and how to report the results of PLS-SEM," *Eur. Bus. Rev.*, vol. 31, no. 1, pp. 2–24, Jan. 2019.
- [128] C. M. Ringle, M. Sarstedt, and D. W. Straub, "Editor's comments: A critical look at the use of PLS-SEM in 'MIS quarterly," *MIS Quart.*, vol. 36, no. 1, pp. 3–14, 2012.
- [129] S. Streukens and S. Leroi-Werelds, "Bootstrapping and PLS-SEM: A step-by-step guide to get more out of your bootstrap results," *Eur. Manage. J.*, vol. 34, no. 6, pp. 618–632, Dec. 2016.
- [130] M. Sarstedt, C. M. Ringle, J.-H. Cheah, H. Ting, O. I. Moisescu, and L. Radomir, "Structural model robustness checks in PLS-SEM," *Tourism Econ.*, vol. 26, no. 4, pp. 531–554, Jun. 2020.
- [131] T. Ramayah, J. Cheah, F. Chuah, H. Ting, and M. A. Memon, "Partial least squares structural equation modeling (PLS-SEM) using smartPLS 3.0," in *An Updated Guide and Practical Guide to Statistical Analysis*. Kuala Lumpur, Malaysia: Pearson, 2018.

- [132] R. B. Kline, Principles and Practice of Structural Equation Modeling. New York, NY, USA: Guilford Publications, 2023.
- [133] H. H. Harman, Modern Factor Analysis, 2nd ed. Chicago, IL, USA: Univ. of Chicago Press, 1976.
- [134] P. M. Podsakoff, S. B. MacKenzie, J.-Y. Lee, and N. P. Podsakoff, "Common method biases in behavioral research: A critical review of the literature and recommended remedies," *J. Appl. Psychol.*, vol. 88, no. 5, pp. 879–903, 2003.
- [135] S. B. MacKenzie and P. M. Podsakoff, "Common method bias in marketing: Causes, mechanisms, and procedural remedies," *J. Retailing*, vol. 88, no. 4, pp. 542–555, Dec. 2012.
- [136] W. W. Chin, "Commentary: Issues and opinion on structural equation modelling," *MIS Quart.*, vol. 22, no. 1, pp. vii–xvi, Mar. 1998.
- [137] C. Fornell and D. F. Larcker, "Evaluating structural equation models with unobservable variables and measurement error," *J. Marketing Res.*, vol. 18, no. 1, pp. 39–50, Feb. 1981.
- [138] J. Henseler, C. M. Ringle, and M. Sarstedt, "A new criterion for assessing discriminant validity in variance-based structural equation modeling," *J. Acad. Marketing Sci.*, vol. 43, no. 1, pp. 115–135, Jan. 2015.
  [139] J. Henseler, G. Hubona, and P. A. Ray, "Using PLS path modeling in
- [139] J. Henseler, G. Hubona, and P. A. Ray, "Using PLS path modeling in new technology research: Updated guidelines," *Ind. Manage. Data Syst.*, vol. 116, no. 1, pp. 2–20, Feb. 2016.
- [140] M. R. A. Hamid, W. Sami, and M. H. M. Sidek, "Discriminant validity assessment: Use of Fornell & Larcker criterion versus HTMT criterion," *J. Phys., Conf. Ser.*, vol. 890, Sep. 2017, Art. no. 012163.
- [141] J. Cohen, "Statistical power analysis," *Current Directions Psychol. Sci.*, vol. 1, no. 3, pp. 98–101, 1992.
- [142] S. S. Sawilowsky, "New effect size rules of thumb," J. Modern Appl. Stat. Methods, vol. 8, no. 2, pp. 597–599, Nov. 2009.
- [143] O. Ciftci, K. Berezina, and M. Kang, "Effect of personal innovativeness on technology adoption in hospitality and tourism: Meta-analysis," in *Proc. ENTER eTourism Conf. Inf. Commun. Technol. Tourism.* Cham, Switzerland: Springer, 2021, pp. 162–174.
- [144] W. R. King and J. He, "A meta-analysis of the technology acceptance model," *Inf. Manage.*, vol. 43, no. 6, pp. 740–755, Sep. 2006.
- [145] A. LeNail, "NN-SVG: Publication-ready neural network architecture schematics," J. Open Source Softw., vol. 4, no. 33, p. 747, Jan. 2019.
- [146] K.-B. Ooi and G. W.-H. Tan, "Mobile technology acceptance model: An investigation using mobile users to explore smartphone credit card," *Expert Syst. Appl.*, vol. 59, pp. 33–46, Oct. 2016.
- [147] V. Andavara, B. Sundaram, D. Bacha, T. Dadi, and P. Karthika, "The impact of perceived ease of use on intention to use mobile payment services for data security applications," in *Proc. 2nd Int. Conf. Electron. Sustain. Commun. Syst. (ICESC)*, Aug. 2021, pp. 1875–1880.
- [148] A. Karahoca, D. Karahoca, and M. Aksöz, "Examining intention to adopt to Internet of Things in healthcare technology products," *Kybernetes*, vol. 47, no. 4, pp. 742–770, Mar. 2018.
- [149] M. Q. M. AlHamad, I. Akour, M. Alshurideh, A. Q. Al-Hamad, B. A. Kurdi, and H. Alzoubi, "Predicting the intention to use Google glass: A comparative approach using machine learning models and PLS-SEM," *Int. J. Data Netw. Sci.*, vol. 5, no. 3, pp. 311–320, 2021.
- [150] M. H. Ronaghi and A. Forouharfar, "A contextualized study of the usage of the Internet of Things (IoTs) in smart farming in a typical middle eastern country within the context of unified theory of acceptance and use of technology model (UTAUT)," *Technol. Soc.*, vol. 63, Nov. 2020, Art. no. 101415.
- [151] P. S. de Boer, A. J. A. M. van Deursen, and T. J. L. van Rompay, "Accepting the Internet-of-Things in our homes: The role of user skills," *Telematics Informat.*, vol. 36, pp. 147–156, Mar. 2019.
- [152] M. Mital, V. Chang, P. Choudhary, A. Papa, and A. K. Pani, "Adoption of Internet of Things in India: A test of competing models using a structured equation modeling approach," *Technol. Forecasting Social Change*, vol. 136, pp. 339–346, Nov. 2018.
- [153] E. Park, Y. Cho, J. Han, and S. J. Kwon, "Comprehensive approaches to user acceptance of Internet of Things in a smart home environment," *IEEE Internet Things J.*, vol. 4, no. 6, pp. 2342–2350, Dec. 2017.
- [154] D. Liu, Q. Li, and S. Han, "Using extended technology acceptance model to assess the adopt intention of a proposed IoT-based health management tool," *Sensors*, vol. 22, no. 16, p. 6092, Aug. 2022.
- [155] E. Lee and S. Han, "Determinants of adoption of mobile health services," Online Inf. Rev., vol. 39, no. 4, pp. 556–573, Aug. 2015.
- [156] A. Alhasan, L. Audah, I. Ibrahim, A. Al-Sharaa, A. S. Al-Ogaili, and J. M. Mohammed, "A case-study to examine doctors' intentions to use IoT healthcare devices in Iraq during COVID-19 pandemic," *Int. J. Pervasive Comput. Commun.*, vol. 18, no. 5, pp. 527–547, Nov. 2022.

- [157] C. L. Miltgen, A. Popovič, and T. Oliveira, "Determinants of end-user acceptance of biometrics: Integrating the 'big 3' of technology acceptance with privacy context," *Decis. Support Syst.*, vol. 56, pp. 103–114, Dec. 2013.
- [158] C. Koo and N. Chung, "Examining the eco-technological knowledge of smart green IT adoption behavior: A self-determination perspective," *Technol. Forecasting Social Change*, vol. 88, pp. 140–155, Oct. 2014.
- [159] J. Hair, J. F. Hair Jr., M. Sarstedt, C. M. Ringle, and S. P. Gudergan, Advanced Issues in Partial Least Squares Structural Equation Modeling. Thousand Oaks, CA, USA: saGe Publications, 2023.
- [160] D. J. Kim, C. Steinfield, and Y.-J. Lai, "Revisiting the role of web assurance seals in business-to-consumer electronic commerce," *Decis. Support Syst.*, vol. 44, no. 4, pp. 1000–1015, Mar. 2008.
- [161] M. Mousavizadeh and D. Kim, "A study of the effect of privacy assurance mechanisms on self-disclosure in social networking sites from the view of protection motivation theory," in *Proc. 36th Int. Conf. Inf. Syst.*, Fort Worth, TX, USA, 2015.
- [162] C. Ye and R. Potter, "The role of habit in post-adoption switching of personal information technologies: An empirical investigation," *Commun. Assoc. Inf. Syst.*, vol. 28, no. 1, p. 35, 2011.
- [163] M. Moslehpour, V. K. Pham, W.-K. Wong, and İ. Bilgiçli, "E-purchase intention of Taiwanese consumers: Sustainable mediation of perceived usefulness and perceived ease of use," *Sustainability*, vol. 10, no. 1, p. 234, Jan. 2018.
- [164] W. R. Lin, C.-Y. Lin, and Y.-H. Ding, "Factors affecting the behavioral intention to adopt mobile payment: An empirical study in Taiwan," *Mathematics*, vol. 8, no. 10, p. 1851, Oct. 2020.
- [165] A. G. Chin, M. A. Harris, and R. Brookshire, "An empirical investigation of intent to adopt mobile payment systems using a trust-based extended valence framework," *Inf. Syst. Frontiers*, vol. 24, no. 1, pp. 329–347, Feb. 2022.



**YUSMADI YAH JUSOH** received the B.Ec. degree in economic analysis and public policy and the master's and Ph.D. degrees in information technology from the National University of Malaysia, in 1997, 1998, and 2008, respectively. She is currently an Associate Professor with the Faculty of Computer Science and Information Technology, Universiti Putra Malaysia. Her current research interests include management information systems, information systems, information

technology strategic planning, and software project management.



**MARZANAH A. JABAR** received the Ph.D. degree in management information systems from Universiti Putra Malaysia (UPM), Malaysia. She is currently an Associate Professor with the Department of Software Engineering and Information Systems, UPM. She has more than 20 years of experience as a system analyst in the area of enterprise system development. She has been appointed as a Consultant to several software development projects at UPM and other agencies.

To date, she is a Principal Investigator of 35 research grants valued at RM2 million and consultation work worth RM300 000. She has published more than 200 articles in journals, conference proceedings, seminars, and technical reports. She has 53 copyrights and two patents. She has also successfully commercialized one product from her research. Her current research interests include software engineering, knowledge management, information management systems, and enterprise software development.



**RADHWAN SNEESL** received the B.Sc. degree in computer science from the University of Basrah, Iraq, and the M.Sc. degree in computer science (artificial intelligence and distributed computing) from the West University of Timisoara, Romania. He is currently pursuing the Ph.D. degree with the Department of Software Engineering and Information Systems, Faculty of Computer Science and Information Technology, Universiti Putra Malaysia. His current research interests include most homes, data englytical information theory.

IoT, smart campuses, smart homes, data analytics, information theory, technology adoption, and the use of qualitative and quantitative methods in information systems research. His career has included management information systems, teaching, research, and analysis.



**SALFARINA ABDULLAH** received the Ph.D. degree in knowledge management from Universiti Putra Malaysia, Malaysia. She is currently a Senior Lecturer with the Department of Software Engineering and Information Systems, Universiti Putra Malaysia. She has served almost 20 years of teaching in the faculty for various undergraduate and postgraduate courses mostly in the area of information systems. She has been appointed as a Consultant and expert subject matter to several

software development projects at UPM and other higher learning institutions. She has led a couple of research grants as a Principal Investigator and a research committee for several other research grants. She has published more than 40 papers in journals and conferences proceedings and seminars. She also has over ten copyrights and one patent. Her current research interests include knowledge management in software engineering and information management systems.